Design and Fabrication of a Post-Processing Furnace for 3D Printed Parts

by

Ram Chilukuri

Bachelor of Technology, Mechanical Engineering Indian Institute of Technology, Delhi, 1998

Submitted to the Department of Mechanical Engineering in Partial Fulfillment of the Requirements for the degree of

MASTER OF SCIENCE IN MECHANICAL ENGINEERING

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

MAY 2000

© 2000 Massachusetts Institute of Technology All rights reserved

Signature of author:	
	Department of Mechanical Engineering
	May 15, 2000
Certified by:	··· ····
	Emanuel M. Sachs
	Professor of Mechanical Engineering Thesis Supervisor
Accepted by:	
	Ain A. Sonin
	Chairman, Graduate Committee
	MASSACHUSETTS INSTITUTE OF TECHNOLOGY SEP 2 0 2000

ENG.

LIBRARIES

Design and Fabrication of a Post-Processing Furnace for 3D Printed Parts

by

Ram Chilukuri

Submitted to the Department of Mechanical Engineering on May 15, 2000 in Partial Fulfillment of the Requirements for the Degree of Master of Science in Mechanical Engineering

ABSTRACT

Three dimensional printing is a rapid prototyping process that forms parts directly from a computer model by joining powdered material with a binder. The green part so obtained is subjected to post processing steps that include debinding, sintering and infiltration. Furnaces currently being used for post processing use a Hydrogen or a vacuum atmosphere and the cost of scaling them up to process large parts is prohibitively high. Previous work done at MIT came up with a design for an economical furnace, and a prototype small scale furnace. In this work, as a first step, the small scale furnace previously developed at MIT was optimized to yield good quality, oxide free sintered and infiltrated parts. The important thing learnt from the small scale furnace shell from the outside and allow the furnace to soak at 200 C until the dew point inside got to a low value like –25 C, before ramping the furnace to 1200 C.

Having learnt about the process and obtained design parameters from the small scale furnace, a large scale furnace was constructed. This furnace has a rectangular cross section and is more than 20 times bigger in volume than the small scale furnace, which had a circular cross section. This furnace has an Aluminum shell, uses Alumina and Graphite insulation and uses Kanthal heating elements embedded in high Alumina content heater plates to heat the inside of the furnace (12 internal heaters for a total of 10.2 kW). It is designed to operate in a Forming gas or Argon atmosphere. In addition to the internal heaters, it has eleven strip heaters on the shell (a total power of 11 kW) to expel moisture from the insulation during the purging phase of a furnace run. The furnace has a Honeywell controller that uses PID control to ensure the desired operation of the furnace. The controller has events that are used to control the switching of the strip heaters and gas solenoid valves. SS420 parts were successfully sintered and infiltrated using the large scale furnace. At 1200 C, excellent temperature uniformity (temperature in a 5 C band) was found to exist inside the furnace both in the vertical direction and along the furnace depth.

A second issue that was studied as part of this work was the effect of directional solidification on porosity in infiltrated parts. The small scale furnace was used to conduct directional solidification experiments on SS420 parts infiltrated with Bronze. These involved inducing a temperature gradient in a part (using a jet of Argon set up to strike the part on the top) at the end of an infiltration cycle. By comparing two extreme cases, directional solidification, in which the top of a solidifying part is kept cooler than the bottom, with reverse directional solidification, in which the bottom of the part is kept cooler than the top as it solidifies, it was established that directional solidification reduces porosity in parts.

Thesis Supervisor: Emanuel M. Sachs Title: Professor of Mechanical Engineering

Acknowledgements

I would like to thank Prof. Emanuel Sachs for giving me a chance to work on this project. I am grateful to Prof. Sachs for all that he taught me over the last two years. I am truly impressed by his ability to come up with simple solutions to seemingly complex problems.

I would like to thank the following people who helped me during the course of my research:

Prof. Samuel Allen, for giving me ideas out of his infinite trove of material science knowledge.

Adam Lorenz, who helped me come up to speed with furnace design. I would like to thank him for being a good colleague, ever ready to use his expertise to help me solve problems.

Julie Baars Kaczynski for promptly processing purchase orders for me, and for all the interesting conversations on a variety of non engineering topics. It was fun working with her.

Diane Brancazio, for designing the furnace shell. It was a pleasure working with her.

David Brancazio and James Serdy, for being ever ready to help me.

Mark Belanger, for making all my visits to the LMP machine shop fun experiences, and for being a good friend. I'd also like to thank Gerry Wentworth for helping me in the machine shop.

Peter Morley of MIT Central Machine Shop, for expediting all my job orders.

Yin- Lin Xie of Material Science, for helping me section, polish and photograph infiltrated parts.

I would also like to thank the following folks for making my stay at the 3D printing lab a memorable one: Andreas Straube, Ben Polito, Berndt Heid, Bjorn DeBear, Costas Hadjiloucas, David Ables, David Guo, Diana Buttz, Enrico Vezzetti, Garth Grover, Huaijun Wu, Markus Werner, Olaf Dambon, Patrick Saxton, Sang-bum Hong, Xiaorong Xu and Zubin Irani. Special thanks to David Guo for being a great friend.

Finally, I would like to thank my parents, C.V. Avadhani and C. Prasanna, for all their love and encouragement, without which this work would not have been possible.

Table of Contents

Abstract	2
Acknowledgements	3
Table of Contents	4
List of Figures	7
List of Tables	

1 Introduction	13
 1.1 Three Dimensional Printing 1.2 Motivation and objective 1.3 Previous work 1.4 Organization of this work 	
2 Insulation Experiments	17
 2.1 Small scale furnace 2.2 Effect of Silica in the heater plates 2.3 Other considerations 2.4 Summary 	
3 Design of a Debinding Apparatus	25
 3.1 Current practice	25 27 28 29 32 33
4 Directional Solidification	
 4.1. Porosity	
4.2.2. Experimental set up	

	4.2.3.	Determination of cooling gas flow rate	38
	4.3. Dir	ectional solidification experiments	41
	4.3.1.	Initial efforts	41
	4.3.2.	Reverse directional solidification	41
	4.3.3.	Modeling surface porosity	43
	4.3.4.	Temperature profiles	45
	4.4. Sur	nmary	49
5	Desigr	and Fabrication of the Large Scale Furnace	50
-	51 Th	ermal design	
	511	Estimating steady state heat losses	
	512	Estimating ramp up and total power requirement	
	513	Selection of internal heaters	
	52 Fu	rnace shell and external features	
	5.2. Tu	External strip heaters	
	527	Fittings on the shell	57
	522	Furnace Doors and heat shields	58
	52.5.	Power connections to internal heaters	59
	5.2.4	vilation	57 62
	5.2.1	Initial affarta	02
	522	Polymor spray	02 64
	5.2.2.	Novtal thread and Inconal mach	04
	524	Initial testing	05
	525	Choice of extra insulation material	60
	5.3.3.		9 73
	J.4. Ju	iiiiiai y	75
6	Electi	rical, Gas and Safety Systems	74
	6.1 Ele	ctrical systems	74
	6.1.1	Power supply	74
	6.1.2	Wiring scheme for external heaters	76
	6.1.3	Controller	79
	6.1.4	Internal heaters	81
	6.1.5	Blower	84
	6.1.6	Gas solenoids	85
	6.2 Gas	systems	85
	6.3 Saf	ety systems	89
	6.4 Cor	nclusion	92
7	Furna	ace Testing	93
	7 1 Ini+	ial furnace runs	02
	7.1 IIII 7.2 Det	armining an optimal furnace run	Q/
	7.2 Del	Furnace runs for obtaining oxide free parts	05
	7.2.1	Other romp schemes	رد ۵۵
	1.2.2	Deve point sensing	101
	1.2.3	Dew point sensing	102
	1.3 IW	o zone control scheme and temperature profiling	102
	1.3.1	Experiments with two zone control	102
	1.3.2	remperature profiling	105

7.4 Summary	
8 Conclusion and Future Work	
8.1 Summary of achievements	
8.2 Recommendations for future work	
 Appendix 1: List of Vendors Appendix 2: Furnace Part Drawings Appendix 3: Sample Program for the Large Scale Furnace Appendix 4: Using a Nextel Lined SS Crucible for Infiltrating Parts References 	

List of Figures

Chapter 1

FIGURE 1.1: SCHEMATIC SHOWING STEPS INVOLVED IN THE THREE DIMENSIONAL PRINTING PROCESS 1	13
FIGURE 1.2: SCHEMATIC SHOWING THE POST PROCESSING STEPS CARRIED OUT ON A GREEN PART	14
FIGURE 1.3: SMALL SCALE FURNACE BUILT AT MIT.	16

Chapter 2

FIGURE 2.1: SCHEMATIC OF THE SMALL SCALE FURNACE SHOWING VARIOUS FEATURES.	17
FIGURE 2.2: PICTURE OF THE SMALL SCALE FURNACE DEVELOPED AT MIT.	18
FIGURE 2.3: DEW POINT VS. TIME: PURGE AT ROOM TEMPERATURE FOR TWO HOURS FOLLOWED BY A RAM	MP
то 1200 С	19
FIGURE 2.4: DEW POINT VS. TIME: SOAK AT 200 C UNTIL DEW POINT GOES TO -25 C, THEN RAMP TO 1200 C	C.
	20
FIGURE 2.5: COMPARISON OF MOISTURE RELEASE AT TEMPERATURES ABOVE 900 C: LOW VS. HIGH	
ALUMINA HEATERS.	21
FIGURE 2.6: PARTS SINTERED IN THE SMALL SCALE FURNACE: (A) WITH NO BAND HEATERS, RAMP TO 120)()
C, FORMING GAS ATMOSPHERE, LOW ALUMINA HEATERS (B) WITH BAND HEATERS, SOAK AT 200 C,	
THEN RAMP TO 1200 C, FORMING GAS ATMOSPHERE, LOW ALUMINA HEATERS (C) SAME AS 2.6B,	
EXCEPT THAT ATMOSPHERE IS SWITCHED FROM FORMING GAS TO ARGON AT 600 C (D) SAME AS 2.61	B,
EXCEPT THAT THE HEATERS IN THE FURNACE HAVE A HIGH ALUMINA CONTENT.	22
FIGURE 2.7: SCHEMATIC OF THE SET UP FOR THE VACUUM TEST CARRIED OUT USING THE SMALL SCALE	
FURNACE	23

Chapter 3

FIGURE 3.1: SCHEMATIC SHOWING THE CONCEPT OF A DEBINDING CHAMBER. NOTE THAT THE ANNULAR	
CLEARANCE BETWEEN THE EFFLUENT TUBE AND THE HOLE IN THE GRAPHITE LID, T =1/2(D-D)	. 25
FIGURE 3.2: SCHEMATIC SHOWING GAS FLOW INTO THE FURNACE AND TWO POSSIBLE EXIT PATHS FOR	
OUTGOING GAS: ONE THROUGH THE ANNULAR GAP AND OUT THROUGH THE EFFLUENT TUBE AND THE	ΗE
OTHER THROUGH THE REGULAR OUTLET O1.	. 26
FIGURE 3.3: SCHEMATIC SHOWING AN INCREASE IN THE EFFECTIVE ANNULAR GAP BETWEEN THE TUBE	
AND THE HOLE DUE TO MISALIGNMENT.	. 30
FIGURE 3.4: PICTURE OF A SS TRUSS THAT WAS SUCCESSFULLY DEBOUND AND SINTERED USING THE	
DESCRIBED SET UP	. 31
FIGURE 3.5: SIMPLIFIED SCHEMATIC OF A MODIFIED FLOW METER. SUCH A FLOW METER, WHEN MOUNT	ED
AT OUTLET O1 OF THE FURNACE CAN BE USED TO MAINTAIN CONSTANT PRESSURE INSIDE THE	
FURNACE DURING THE ENTIRE FURNACE CYCLE WITHOUT USER INTERVENTION	. 32

FIGURE 4.1: SCHEMATIC SHOWING IMPORTANT DIMENSIONS OF THE TEST PART.	
FIGURE 4.2: ILLUSTRATION OF THE SPLIT MOLD CONCEPT: PRE SINTERED METAL CHUNKS ON T	HE OUTSIDE
HOLD THE TWO HALVES OF THE MOLD TOGETHER.	
FIGURE 4.3: (A) SCHEMATIC SHOWING DIMENSIONS CHOSEN FOR THE TEST PART (LEFT) AND, (H	3) PICTURE
OF THE TEST PART (RIGHT).	
FIGURE 4.4: HEAT AND GAS FLOW DIRECTIONS, ASSUMING ONE DIMENSIONAL HEAT TRANSFER	
FIGURE 4.5: PICTURES A, B AND C ARE FROM THE CROSS SECTION OF A DIRECTIONALLY SOLIDI	FIED PART.
PICTURED D, E AND F ARE FROM A PART THAT WAS SOLIDIFIED USING REVERSE DIRECTION	NAL

SPOTS
FIGURE 4.6: SCHEMATIC SHOWING A SHRINKAGE CAVITY FORMED NEAR THE SURFACE OF A COOLING PART. IF THE PRESSURE DIFFERENCE ACROSS THE INFILTRANT FILM AT THE SURFACE IS TOO HIGH, THE FILM MIGHT RUPTURE GIVING RISE TO SURFACE POROSITY
PART. IF THE PRESSURE DIFFERENCE ACROSS THE INFILTRANT FILM AT THE SURFACE IS TOO HIGH, THE FILM MIGHT RUPTURE GIVING RISE TO SURFACE POROSITY
THE FILM MIGHT RUPTURE GIVING RISE TO SURFACE POROSITY.
FIGURE 4.7: APPROXIMATION SHOWING THE CROSS SECTION OF A CHANNEL THAT THE LIQUID INFILTRANT
COULD FLOW THROUGH TO REACH A SHRINKAGE CAVITY AND FILL IT
FIGURE 4.8: SCHEMATIC SHOWING THE TWO HOLES THAT WERE DRILLED INTO A PART TO DO
TEMPERATURE PROFILING INSIDE THE PART
FIGURE 4.9: CROSS SECTION OF MULLITE SHEATH USED FOR THE TYPE K PROFILING THERMOCOUPLE 46
FIGURE 4.10: PICTURE OF THE SET UP USED FOR TEMPERATURE PROFILING. NOTE THE MULLITE SHEATH
AND THE ALUMINUM CLAMPS
FIGURE 4.11: GRAPH SHOWING THE VARIATION OF TEMPERATURE INSIDE AN INFILTRATED PART WITH AND
WITHOUT COOLING GAS. THE COOLING GAS INDUCES A GRADIENT OF AROUND 5 C/CM IN THE PART.
THE 1200 C IN THE LEGEND INDICATES THE FURNACE THERMOCOUPLE READING IN STEADY STATE 48
FIGURE 4.12: TEMPERATURE PROFILE MEASURED IN A COOLING PART. A POSITIVE GRADIENT INDICATES
THAT THE TOP IS COOLER THAN THE BOTTOM. NOTE THE EFFECT THE COOLING GAS FLOW HAS ON THE
GRADIENT. THE SHARPLY INCREASING TEMPERATURE GRADIENT SEEN AT T=0 IS A TRANSIENT
CAUSED BY TURNING THE COOLING GAS ON 15 SEC BEFORE T=0. ALSO SHOWN IN THE GRAPH IS THE
FURNACE TEMPERATURE

FIGURE 5.1: SCHEMATIC SHOWING DIMENSIONS OF THE HOT FACES OF THE INSULATION THAT GOES INTO
THE SMALL SCALE FURNACE. THESE DIMENSIONS WERE USED TO CALCULATE THE AREA OF THE HOT
ZONE OF THE FURNACE
FIGURE 5.2: RESISTANCE TO HEAT TRANSFER FROM THE HOT ZONE THAT IS AT 1200 C TO THE AMBIENT
ATMOSPHERE. HEAT TRANSFER INVOLVES CONDUCTION THROUGH THE INSULATION, CONDUCTION
THROUGH THE ALUMINUM SHELL AND CONVECTION TO THE ATMOSPHERE
FIGURE 5.3: SCHEMATIC SHOWING DEAD SPACE BETWEEN THE PART PISTON AND THE INSULATION.
DIMENSIONS OF THE INSULATION WERE ESTIMATED BY ASSUMING VALUES FOR THE DEAD SPACE
BETWEEN THE PISTON AND THE HOT FACE OF THE INSULATION. THE DIMENSIONS THUS OBTAINED
WERE USED TO DETERMINE STEADY STATE HEAT LOSSES BY SCALING UP THOSE FROM THE SMALL
SCALE FURNACE BASED ON THE AREA OF INSULATION EXPOSED TO 1200 C.
FIGURE 5.4: PICTURE OF THE LARGE SCALE FURNACE. NOTE THE STRIP HEATER AND THE VIEW PORTS ON
THE FRONT DOOR
FIGURE 5.5: PICTURE SHOWING A TOP VIEW OF THE FURNACE SHELL. NOTE THE QF40 FITTINGS AND THE
THERMOCOUPLE FEEDTHROUGHS
FIGURE 5.6: COMPARISON OF THE TWO KINDS OF DOORS THAT CAN BE USED WITH THE FURNACE. THE LEFT
DOOR IS FOR USE DURING TEMPERATURE PROFILING AND THE RIGHT DOOR WITH THE VIEWPORTS IS
FOR USE DURING REGULAR FURNACE RUNS
FIGURE 5.7: PICTURE OF THE FURNACE FITTED WITH A DOOR USED FOR TEMPERATURE PROFILING. THE
DOOR HAS SIX PORTS FOR INSERTING THERMOCOUPLES INTO THE FURNACE. THE THERMOCOUPLES
CAN BE USED TO MEASURE THE TEMPERATURE INSIDE THE FURNACE AS A FUNCTION OF THE
DISTANCE FROM THE DOOR
FIGURE 5.8: PICTURE SHOWING AN INTERNAL HEATER WITH ITS LEADS BRAZE JOINED TO BRASS TUBES.
THE HEATER LEADS STICK 1/2" INTO THE BRASS TUBES. THE BRAZE IS A SILVER-COPPER EUTECTIC 60
FIGURE 5.9: PICTURE SHOWING THE BRAZE JOINT BETWEEN THE HEATING ELEMENT AND THE BRASS
TUBE. ONCE OUTSIDE THE HEATER PLATE, THE HEATING ELEMENT IS DOUBLE TWISTED TO REDUCE
RESISTIVE HEATING. THIS KEEPS THE BRAZE JOINT COOL60
FIGURE 5.10: PICTURE SHOWING THE SWAGELOK FITTING AND TEFLON SLEEVE ASSEMBLY THAT WAS
TESTED AS A POSSIBLE CANDIDATE FOR USE AS A POWER FEEDTHROUGH
FIGURE 5.11: PICTURE SHOWING CREEP IN THE TEFLON SLEEVE THAT WAS USED IN A SWAGELOK FITTING
AND MAINTAINED AT 200 C FOR A FEW HOURS
FIGURE 5.12: ULTEM FITTING WHICH WAS USED AS A POWER FEEDTHROUGH FOR HEATER WIRES
FIGURE 5.13: PICTURE DEMONSTRATING HOW A SILICONE TUBE CAN BE USED TO ACHIEVE A LEAK PROOF
JOINT AT THE ULTEM FITTING-BRASS TUBE INTERFACE

FIGURE 5.14: PICTURE SHOWING THE POWER FEEDTHROUGH ASSEMBLY. THE WIRE FROM THE SCR IS
CONNECTED TO THE BRASS TUBE BY A BRASS CONNECTOR. THE ULTEM FITTING IS SCREWED INTO
THE ALUMINUM STANDOFF
FIGURE 5.15: DRAWING OF THE BRASS CONNECTOR USED TO CONNECT A WIRE FROM THE SCR TO A BRASS
TUBE THAT IS BRAZED TO A HEATER LEAD. HOLES A AND B IN THE DRAWING ARE CLEARANCE HOLES
FOR 8-32 AND 6-32 SCREWS RESPECTIVELY63
FIGURE 5.16: SCHEMATIC SHOWING A SECTION THROUGH AN ALUMINUM STANDOFF ON THE FURNACE
WALL. THE MULLITE TUBE INSIDE THE STANDOFF IS TO PREVENT THE HEATER WIRE FROM TOUCHING
THE ALUMINUM SHELL
FIGURE 5.17: PICTURE OF THE RESULTS OF FIRING INSULATION SPRAYED WITH RHOPLEX TO 1200 C. THE
INSULATION ON THE LEFT WAS RAMPED TO 1200 C AT 5 C/MIN STARTING AT ROOM TEMP., WHILE
THAT ON THE RIGHT WAS HELD IN THE FURNACE AT 500 C FOR AN HOUR BEFORE RAMPING TO 1200 C
AT 5 C/MIN. THE IDEA WAS TO GIVE TIME FOR COMPLETE POLYMER BURNOUT AT 500 C BEFORE
RAMPING TO 1200 C. BOTH THESE RESULTS WERE UNACCEPTABLE
FIGURE 5.18: PICTURE SHOWING ALUMINA INSULATION SEWN TOGETHER ALONG THE PERIPHERY USING
NEXTEL THREAD
FIGURE 5.19: INTERIOR OF FURNACE. INCONEL WIRE MESH SUPPORTS THE INSULATION ON THE TOP, BACK,
AND FRONT FACES OF THE FURNACE. NEXTEL SLEEVE ON THE PERIPHERY OF THE FRONT INSULATION
PROTECTS IT FROM BEING DAMAGED WHEN THE FURNACE IS BEING LOADED/UNLOADED. ALSO NOTE
HOW THE HEATERS ARE MOUNTED ON THE INCONEL SUPPORTS.
FIGURE 5.20: SCHEMATIC SHOWING A SECTION THROUGH THE ENCLOSURE THAT WAS BUILT AROUND A
SINGLE HEATER. THIS HEATER WAS POWERED BY A VARIAC AND THE VOLTAGE REQUIRED TO ATTAIN
A STEADY STATE TEMPERATURE OF 1200 C INSIDE THE ENCLOSURE WAS FOUND AS A FUNCTION OF
THE QUANTITY OF INSULATION ON THE 9" BY 7" FACE OPPOSITE THE HEATER PLATE
FIGURE 5.21: DEW POINT INSIDE THE FURNACE VS. TIME FOR A SAMPLE OF ALUMINA INSULATION FIRED IN
THE TUBE FURNACE. NOTE THE HUMPS IN DEW POINT WHEN THE FURNACE TEMP. IS 400 AND 600 C. 70
FIGURE 5.22: DEW PT. VS. TIME FOR THE EMPTY FURNACE. THIS IS A CONTROL RUN. NOTE THE ABSENCE
OF HUMPS IN DEW POINT UNLIKE IN FIGURE 5.2171
FIGURE 5.23: DEW PT. VS. TIME FOR A RUN WITH GRAPHITE INSIDE THE FURNACE. NOTE THAT THE
AVERAGE DEW POINT LEVEL IS FAR LESS THAN THAT FOR ALUMINA (FIGURE 5.21). ALSO, HUMPS IN
DEW POINT ARE NOT AS PROMINENT AS IN FIGURE 5.21
FIGURE 5.24: PICTURE SHOWING THE ¼" GRAPHITE INSULATION THAT GOES BEHIND THE 1" ALUMINA
INSULATION. ALSO SEEN IN FRONT OF THE ALUMINA INSULATION ON THE DOOR IS A SECOND LAYER
OF ¼" GRAPHITE INSULATION, WHOSE PURPOSE IS TO PREVENT ANY LEAKING HOT GAS FROM
REACHING THE ALUMINUM SHELL AND TO ACT AS A GETTERING AGENT

FIGURE 6.1: CONVENTION USED FOR REPRESENTING WIRES IN ALL WIRING DIAGRAMS IN THIS CHAPTER. 74
FIGURE 6.2: SCHEMATIC SHOWING WIRING FROM THE 240 V AC SOURCE TO THE 120 V AC BUS USED TO
POWER THE CONTROLS OF THE FURNACE. ALSO SHOWN IN THE FIGURE IS THE WIRING OF THE BLOWER
THAT SHALL BE DISCUSSED IN SECTION 6.1.5
FIGURE 6.3: FIGURE SHOWING WIRING OF THE 24 V DC POWER SUPPLY (ON THE LEFT). ALSO SHOWN IN
THE FIGURE ON THE RIGHT IS A SCHEMATIC OF THE CONTROLLER PINS, TO BE DISCUSSED LATER 76
FIGURE 6.4: WIRING DIAGRAM FOR THE 24 V AC POWER SUPPLIES THAT ARE USED TO POWER THE SCR
FIRING CARDS, WHOSE WIRING SCHEMATIC IS SHOWN ON THE RIGHT. THE FIRING CARDS ARE
PHYSICALLY MOUNTED ON TO THE SCRS
FIGURE 6.5: WIRING DIAGRAM FOR RELAY R1 THAT TURNS BANK 1 OF EXTERNAL HEATERS ON/OFF. ALSO
SEE FIGURE 6.7
FIGURE 6.6: WIRING DIAGRAM FOR RELAY R2 THAT TURNS BANK 2 OF EXTERNAL HEATERS ON/OFF. ALSO
SEE FIGURE 6.7
FIGURE 6.7: WIRING DIAGRAM FOR THE ELEVEN EXTERNAL HEATERS. ALSO SEE FIGURE 6.5 AND FIGURE
6.6
FIGURE 6.8: SCHEMATIC SHOWING RELATIVE POSITIONS OF THE INTERNAL HEATER PLATES AND POWER
CONNECTIONS TO EACH PLATE. SEE FIGURE 6.9 FOR DETAILS ON HOW THE HEATERS ARE HOOKED TO
THE SCRS

FIGURE 6.9: SCHEMATIC OF WIRING BETWEEN THE SCRS AND THE HEATERS. ALSO SEEN IN THE FIGURE
ARE THE CONTROL SIGNALS TO THE SCRS. SEE FIGURE 6.10 FOR SCHEMATIC OF WIRING FROM 240 V
AC SUPPLY TO SCRS
FIGURE 6.10: WIRING FROM THE 240 V AC SUPPLY TO THE SCRS. THE CONTACTOR IS TURNED ON/OFF BY A
120 V AC SIGNAL
FIGURE 6.11: SCHEMATIC OF THE WIRING OF THE POTENTIOMETERS THAT CAN BE USED TO TRIM THE
VOLTAGE SIGNAL GOING TO THE SCRS. THE CONTROLLER GIVES A 4-20 MA SIGNAL THAT IS
CONVERTED TO A VOLTAGE SIGNAL USING THE RESISTOR R.
FIGURE 6.12: SCHEMATIC OF ALL THE RESISTANCES IN PARALLEL WITH EACH OTHER AS SEEN FROM THE
CONTROLLER'S OUTPUT TERMINALS 43 AND 44. THE IDEA IS TO DETERMINE R THAT WILL CAUSE THE
EQUIVALENT RESISTANCE ACROSS PINS 43 AND 44 TO BE 0.3 K.
FIGURE 6.13: WIRING DIAGRAM FOR RELAY 356CR WHICH IS USED TO CLOSE (OPEN) THE NO ARGON
SOLENOID WHILE SIMULTANEOUSLY OPENING (CLOSING) THE NC FORMING GAS SOLENOID
FIGURE 6.14: SCHEMATIC OF THE GAS SYSTEM USED IN THE FURNACE. THERE ARE TWO POSSIBLE
ATMOSPHERES THE FURNACE HAS BEEN SET UP TO RUN IN: ARGON AND FORMING GAS
FIGURE 6.15: SCHEMATIC OF THE GAS FLOW LINE FROM THE FURNACE OUTLET TO THE EXHAUST DUCT 89
FIGURE 6.16: SCHEMATIC OF THE WIRING OF SAFETY SWITCHES IN SERIES. THE SWITCHES CARRY THE 120
V AC CONTROL SIGNAL TO THE CONTACTOR, 258CON, THAT TURNS ON POWER TO THE INTERNAL
HEATERS
FIGURE 6.17: SCHEMATIC OF THE FLOW ALARM'S PINS. SWITCH S3 IS NORMALLY CLOSED. THE FLOW
ALARM SENSOR BODY IS MOUNTED ON THE FLOW METER, WHILE THE CIRCUIT BOARD IS MOUNTED ON
THE FURNACE FRAME
FIGURE 6.18: WIRING DIAGRAM FOR THE OVERTEMP. BUZZER ALARM WHICH GETS ACTIVATED BY AN
ALARM SIGNAL FROM THE OVERTEMP. CONTROLLER. THE SPRING LOADED PUSHBUTTON, WHEN
PRESSED, SILENCES THE ALARM

FIGURE 7.1: GRAPH SHOWING THE RESPONSE OF THE FURNACE TO A STEP CHANGE IN SET POINT FROM
1180 C TO 1200 C. THE ABSENCE OF ANY SIGNIFICANT OVERSHOOT IN THE RESPONSE SHOWS GOOD
CONTROLLER ACTION
FIGURE 7.2: GRAPH SHOWING THE RESPONSE OF THE FURNACE TO A STEP CHANGE IN SET POINT FROM 800
C TO 780 C. THE ABSENCE OF ANY SIGNIFICANT OVERSHOOT IN THE RESPONSE SHOWS GOOD
CONTROLLER ACTION
FIGURE 7.3: DEW POINT CURVE FOR A FURNACE RUN (RUN 1) CONDUCTED USING THE SCHEME DESCRIBED
IN THE TEXT. NOTE THAT THE PURGING TIME (REGIONS 1 AND 2) IS AROUND 4 HOURS. COMPARE THIS
WITH RUN 2 IN FIGURE 7.5, WHERE THE PURGE TIME IS 3 HOURS
FIGURE 7.4: PICTURE OF THE SS420 PART THAT WAS SUCCESSFULLY SINTERED IN THE FURNACE USING THE
SCHEME DESCRIBED IN SECTION 7.2.1. THE DEW POINT CURVE FOR THIS RUN IS SHOWN IN FIGURE 7.3.
FIGURE 7.5: DEW POINT CURVE FOR RUN 2, CONDUCTED RIGHT AFTER RUN 1 WHOSE DEW POINT CURVE IS
SHOWN IN FIGURE 7.3. THE FURNACE WAS OPENED FOR 40 MINUTES AFTER IT COOLED DOWN TO
ROOM TEMPERATURE AT THE END OF RUN 1, BEFORE STARTING RUN 2. NOTE THAT THE PURGING TIME
FOR THIS RUN IS AROUND 3 HOURS AS OPPOSED TO 4 HOURS FOR RUN 1
FIGURE 7.6: DEW POINT CURVE FOR RUN 5. THE BOTTOM ZONE OF INTERNAL HEATERS WAS KEPT ON WITH
A SET POINT OF 200 C DURING THE PURGE IN ORDER TO QUICKLY HEAT UP THE INSIDE AND SPEED UP
THE MOISTURE REMOVAL PROCESS
FIGURE 7.7: DEW POINT CURVE FOR RUN 6. THE GAS FLOW RATE IN THIS RUN WAS 30 SCFH AS OPPOSED TO
THE 13 SCFH USED IN RUN 5. NOTE THAT THE PURGE TIME IN THIS RUN IS 122 MINUTES, MORE THAN
60 MINUTES LOWER THAN THAT IN RUN 5, WHOSE DEW POINT CURVE IS SHOWN IN FIGURE 7.6 100
FIGURE 7.8: PICTURE OF A SS 420- COPPER PART THAT WAS INFILTRATED SUCCESSFULLY DURING RUN 6.
THE DEW POINT CURVE OF THE CYCLE USED IN RUN 6 IS SHOWN IN FIGURE 7.7. THE PART WAS
SUSPENDED INSIDE THE FURNACE USING THE 1/8" DIAMETER MOLY. ROD SEEN IN THE PICTURE, AND
DIPPED INTO THE COPPER MELT ONLY AFTER ALL THE COPPER HAD MELTED AND STEADY STATE HAD
BEEN ATTAINED AT 1200 C
FIGURE 7.9: PICTURE OF THE FURNACE FITTED WITH THE DOOR USED FOR TEMPERATURE PROFILING 103

FIGURE 7.10: SCHEMATIC OF A SECTION THROUGH THE FURNACE, AS SEEN FROM THE SIDE. PROFILING
WAS DONE FOR X=0 TO X=18.5" ALONG THE FURNACE DEPTH AT EACH OF THE SIX PORTS SHOWN IN
FIGURE 7.9. THE FRONT, MIDDLE AND REAR HEATER PLATES ARE RESPECTIVELY 6", 8" AND 6" LONG
AND ROUGHLY EXTEND FROM THE FRONT INSULATION TO THE REAR INSULATION 104
FIGURE 7.11: SCHEMATIC SHOWING THE RELATIVE LOCATION OF THE THERMOCOUPLE PORTS ON THE
FRONT DOOR
FIGURE 7.12: TEMPERATURE PROFILES OBTAINED ALONG THE DEPTH OF THE FURNACE AT EACH OF THE
SIX PORTS SHOWN IN FIGURE 7.11. NOTE THAT THE TEMPERATURE IS IN A 5 C BAND OVER THE
LENGTH OF THE HEARTH PLATE 106
FIGURE 7.13: TEMPERATURE PROFILES OBTAINED AT PORT 2 AS PART OF THE TRIMMING EXPERIMENT
CARRIED OUT TO TRY AND SMOOTH OUT THE TEMPERATURE PROFILE INSIDE THE FURNACE
FIGURE 7.14: TEMPERATURE PROFILE ALONG THE HEIGHT OF THE FURNACE, MEASURED USING A PORT IN
THE MIDDLE OF THE TOP FACE OF THE FURNACE. NOTE THE TIGHT 2 C BAND THE TEMPERATURE IS IN
OVER A 14" HEIGHT 108

List of Tables

Chapter 2

TABLE 5.1: ESTIMATION OF THE THERMAL MASS OF A FULLY LOADED FURNACE	. 53
THE BELLED THE THE THERE IN NOT THE ENDED TO THE COMPLET FOR THE PARTY IN THE PARTY INTERPARTY IN THE PARTY INTERPARTY IN THE PARTY IN THE PARTY INTERPARTY INTERPARTY INTERPARTY INTERPARTY INTERPARTY	

1 Introduction

1.1 Three Dimensional Printing

Three dimensional printing (3DP) is a rapid prototyping process that forms parts directly from a computer model by joining powdered material with a binder. A thin layer of powder is spread and is selectively joined by a binder deposited using ink jet printing technology. The powder bed is then lowered, another powder layer is spread and binder is deposited to print the next cross section. This process is repeated until the whole part is generated. Figure 1.1 shows the steps involved in 3DP. The part thus obtained is called a green part. 3DP is fast and flexible, allowing the use of different material and binder systems.





The green part is delicate and must undergo further processing before it can be used. The green part has 60% metal powder by volume, and the rest is void space. The binder is burned out in a furnace where the part also undergoes light sintering. The part is then sintered and infiltrated with a low melting point alloy to fill the porosity. Figure 1.2 shows the post processing steps.





1.2 Motivation and objective

Existing furnaces such as the MRF furnace currently being used to post process 3D printed parts at MIT are expensive. The MRF furnace is capable of holding a vacuum and running in a pure hydrogen atmosphere. Scaling up this furnace would be very expensive since the cost of components such as heating elements, heat shields, vacuum pumps, and shell wall capable of withstanding vacuum increase rapidly with size.

The primary focus of this research is on developing a post-processing furnace that can be scaled up economically.

Interconnected porosity in 3D printed parts is a phenomenon that needs to be understood and eventually eliminated. Porosity that is connected is not desirable since it causes cooling channels in 3D printed injection molding tooling to leak. It is believed that porosity occurs due to solidification shrinkage, trapped gases and evolution of gases from the infiltrant melt during solidification. Previous work has tried to investigate some of these issues: however, to date, leak free molds have not been obtained. Lack of a custom furnace has impeded a comprehensive investigation of the porosity issue. One of the objectives of this work is to develop a custom furnace and equip it with features needed to conduct experiments to identify and control parameters affecting porosity. Experiments include gated infiltration and directional solidification. Gated infiltration refers to separating the molten infiltrant from the sintered skeleton until a gate is opened. Directional solidification involves inducing a temperature gradient in a cooling part such that infiltrant in different zones of the part solidifies at different points in time.

Currently, MIT uses two separate MRF furnaces for post processing: one for debinding the green part and the other for sintering and infiltrating it. The furnace used for debinding gets contaminated with condensed vapors of the polymeric binder thus making it unsuitable to be used for sintering or infiltration. Hence, a separate furnace is needed for sintering and infiltration. Great savings in capital cost could be achieved if one furnace could be used for all three post-processing steps. This work also develops a method by which the same furnace can be used for all the post processing steps.

1.3 Previous work

Previous work on scalable furnace processing was done at the MIT 3D printing lab by Adam Lorenz [4]. Lorenz built and tested a series of experimental furnaces in order to come up with a design for an economical furnace. The design consisted of a gas tight outer shell, Alumina insulation and metallic-heating elements embedded in ceramic heater plates. It used a forming gas atmosphere instead of pure Hydrogen or a vacuum in order to achieve significant cost savings. MIT then hired a furnace manufacturing company, C.I. Hayes of Providence, RI, to build a large scale furnace. However, this furnace failed to provide a proper atmosphere for sintering (resulting in oxidized parts), necessitating a review of some of the design elements. The furnace built by Hayes used 3" of Alumina board as insulation. It was found that the board released water vapor even when the furnace was at 1200 C. This was because insulation away from the shell did not get hot enough to expel moisture even when the inside of the furnace was at 1200 C. Moreover, the board had silica-based binder that decomposed and contaminated the furnace atmosphere at high temperatures. As part of a revised design, the board was replaced by high Alumina content (95%) blanket insulation. In order to expel moisture out of the insulation more effectively, it was decided that the shell might need to be heated from the outside. In order for the outside heating to be effective, the shell material was changed from SS to Aluminum, a better thermal conductor. These design changes were incorporated in a small scale prototype furnace consisting of an aluminum shell, alumina insulation and Kanthal heaters (Figure 1.3).



Figure 1.3: Small scale furnace built at MIT.

This research is aimed at optimizing the small scale furnace to yield good quality parts, and using the parameters thus obtained to build and test a large scale furnace, which in addition to demonstrating scalability could be used to conduct experiments on porosity.

1.4 Organization of this work

This work is organized into seven chapters. The second chapter deals with experiments conducted on Alumina insulation to understand its moisture release properties as a function of temperature. It also explains how the small scale furnace was optimized to obtain oxide free parts. Chapter three explains the design of a debinding mechanism that was successfully built and tested in the small scale furnace. Chapter four describes experiments on directional solidification conducted mainly on the SS420-Bronze material system using the small scale furnace. Chapter five deals with the mechanical and thermal system design of the large scale furnace. A lot of the design parameters used were obtained experimentally. Chapter six deals with the electrical, gas and safety systems of the large scale furnace. It gives a detailed description of the control and wiring scheme of the furnace. Chapter seven describes furnace testing and optimization. Finally, chapter eight summarizes this work and lays ground for future work using the furnace built in this work.

2 Insulation Experiments

This chapter describes how a previously developed small scale furnace was optimized to yield oxide free parts. Doing this involved a series of experiments conducted to study the moisture content inside the furnace as a function of temperature.

2.1 Small scale furnace





The small scale furnace (Figures 2.1 and 2.2) was designed to have an operating temperature of 1200 C on the inside and a shell temperature of 200 C in steady state. An inch of high purity (95%) Alumina insulation was used to achieve this. The furnace would run in Forming gas (95% Argon and 5% Hydrogen) and a volume flow rate of five volume changes an hour was considered enough to prevent oxidation of stainless steel (SS) parts.

A typical sintering run involves soaking the SS part at 1200 C for an hour. When the furnace was ramped to 1200 C at 5 C/min starting at room temperature, it was found that the parts got oxidized during sintering.



Figure 2.2: Picture of the small scale furnace developed at MIT.

This was because moisture was being released from the hygroscopic Alumina insulation even when the furnace was at 1200 C, at which temperature SS was most susceptible to oxidation. In order to overcome this problem, the furnace was purged at room temperature for two hours before ramp up until the dew point reached a sufficiently low value (-25 C). However, as soon as ramp up started, the dew point shot up to high levels (Figure 2.3). Clearly, moisture was being expelled out of the insulation as the insulation got heated up. Also, the outer parts of the insulation did not reach 100 C or above until the inside was around 1000 C. Thus water was continued to be expelled even when the furnace temperature was in a region where steel was most susceptible to oxidation (the reactivity follows an Arrhenius temperature dependence). No doubt, the part was oxidized (Figure 2.6a). In order to overcome this problem, two band heaters were mounted around the shell of the furnace. These were provided with thermostats to control the shell temperature at 200 C. The idea was to turn the internal heaters and the external heaters on at a set point of 200 C and purge the furnace with forming gas at five volume changes an hour until the dew point got to -25 C. This would ensure that all the insulation soaked at 200 C and thus any water in the insulation would be expelled before the ramp to 1200 C started. This method worked and produced a part that was much cleaner than that produced in the previous run (Figure 2.6b). Figure 2.4 shows the dew point as a function of time in one such run. It is interesting to note that the dew point starts going up when the ramp begins and peaks when the furnace is at around 480 C before starting to down again. Another interesting fact is that the dew point begins to rise as the furnace temperature goes past 900 C. This behavior can also be seen in figure 2.3. Since the entire vapor from the Alumina insulation had been expelled by this time, some other phenomenon was taking place at furnace temperatures above 900 C.



Figure 2.3: Dew point vs. time: purge at room temperature for two hours followed by a ramp to 1200 C.

2.2 Effect of Silica in the heater plates

Clearly, a chemical reaction was taking place at temperatures above 900 C. A closer look at the heater plate composition showed that the plates had 33% Silica. It was hypothesized that a possible cause of the water vapor could be the reduction of this Silica by Hydrogen present in the Forming gas. In order to test this hypothesis, a run was carried out in which the atmosphere was switched from Forming gas to Argon. This switch was done at 600 C so that at a ramp rate of 5 C/min and a flow rate of 5 volume changes an hour, the entire furnace would be purged of Hydrogen by the time the furnace got to 900 C. If a chemical reaction involving Hydrogen was causing a rise in dew point (and oxidation of the part) at temperatures above 900 C, then, the switch to Argon should yield a clean part (since there would be no Hydrogen to reduce the Silica and produce water vapor). Indeed, the part quality obtained (Figure 2.6c) was much better than



Figure 2.4: Dew point vs. time: soak at 200 C until dew point goes to -25 C, then ramp to 1200 C.

that obtained in the run without the switch.

With sufficient reason to believe that Silica in the heaters was responsible for the rise in dew point level at furnace temperatures above 900 C, a set of new heaters having 93% Alumina and just 7% Silica as opposed to the 33% in the previous heaters was installed in the furnace. Figure 2.5 compares the variation of dew point with time when the furnace was run with the low Alumina heaters and the high Alumina heaters respectively. It can be seen that the rise in dew point when the furnace temperature is around 900 C is negligible with the high Alumina heaters as compared to that seen with the low Alumina heaters. Also, the part obtained had no oxidation when the high Alumina heaters were used (Figure 2.6d). This can be explained by the fact that

the heaters had low Silica content and thus water vapor released by the reduction of Silica was minimal.

Thus, it was established that the reduction of Silica present in the heater plates was indeed responsible for the increase in the amount of water vapor in the furnace at high temperatures. Also, it was found that as the furnace was run in Forming gas, over time, the Silica got continuously depleted and the rise in dew point at high furnace temperatures was no longer seen.



Figure 2.5: Comparison of moisture release at temperatures above 900 C: low vs. high Alumina heaters.

2.3 Other considerations

In order to ensure that the parts sintered in the furnace were oxide free, a moisture trap was used in the gas flow line. Also, high purity Forming gas having less than 10 PPM of water vapor was used as opposed to the previously used industrial grade Forming gas that had up to 30 PPM of moisture. It was decided to replace the Alumina insulation on the top door of the furnace with Graphite. Graphite would act as a gettering agent by reacting with and removing any water

vapor in the furnace. If the dew point inside the furnace is relatively high over several runs (as a result of say, not purging the furnace long enough at 200 C before ramping up to 1200 C), a



Figure 2.6: Parts sintered in the small scale furnace: (a) with no band heaters, ramp to 1200 C, Forming gas atmosphere, low Alumina heaters (b) with band heaters, soak at 200 C, then ramp to 1200 C, Forming gas atmosphere, low Alumina heaters (c) same as 2.6b, except that atmosphere is switched from Forming gas to Argon at 600 C (d) same as 2.6b, except that the heaters in the furnace have a high Alumina content.

visible change can be seen in the Graphite, indicating loss of material by oxidation. The chapter on the assembly of the large scale furnace discusses the issue of Graphite vs. Alumina in greater detail.

A vacuum test was performed on the furnace to get an idea of the size of any leaks that existed in the furnace. Figure 2.7 shows a schematic of the set up for the vacuum test. The idea was to seal all the furnace openings, pull a vacuum inside the furnace and observe the rate of pressure rise inside the furnace. This could then be used to compute the average gas flow rate into the furnace through the leaks. In order to dilute the effect of inflow through the leaks, purging gas a twenty times the leak flow rate could be made to flow through the furnace. It is important to note that this method is extremely conservative since the leak rate into the furnace when it is running with positive pressure inside will be much less than the leak rate when there is a vacuum inside. All openings in the furnace were sealed off and a vacuum was pulled inside the furnace. The pump was shut off and the pressure rise in the furnace was recorded as a function of time. Table 2.1 shows the rise in furnace pressure as a function of time (t = 0 is when the vacuum is pulled).



Figure 2.7: Schematic of the set up for the vacuum test carried out using the small scale furnace.

The volume inside the furnace, neglecting that of the tubes going to the pump and the gauge is V = 0.006613 cubic meters. The air in the room was at a temperature T = 24 C. A pressure change inside the furnace, $\Delta p = 5$ Torr corresponds to a difference in density $\Delta \rho = \Delta p/RT$, where R is the gas constant. The mass of air that leaked in is $\Delta m = \Delta \rho * V$, which comes to $5.17*10^{-5}$ kg. The average flow rate of air can be calculated from the times in the table to be about $5*10^{-7}$ kg/sec. That gives a ball park value of the air leakage that can be expected. In order to dilute this

Time, seconds	Pressure, Torr
0	10
105	15
220	20
330	25

 Table 2.1: Furnace pressure vs. time. A vacuum was pulled into a "sealed" furnace and the pressure rise as a function of time was used to quantify leaks into the furnace.

leakage, we may want the flow rate of forming gas to be 20 times this value, i.e. around $1*10^{5}$ kg/sec. The molecular weight of forming gas, which is 95% Argon and 5% hydrogen by volume is 0.95*40+0.05*2 = 38.1 g/mole. This means that the gas constant for Forming gas is 8314/38.1 = 218.2 kJ/kg K (8314 J/kg K is the universal gas constant). The density of forming gas is p/RT= $10^{5}/(218.2*298) = 1.54$ kg/cubic meter. Thus a mass flow of $1*10^{-5}$ kg/sec corresponds to a flow rate of $1*10^{-5}/1.54 = 0.64*10^{-5}$ cubic meter/sec. This is about 4 volume changes an hour. Thus, the flow rate of five volume changes an hour used in the experiments described in this chapter is a good flow rate.

2.4 Summary

It was found from experiments that the following were needed to obtain good quality, and unoxidized sintered parts in the small scale furnace:

- external heaters to keep the aluminum shell at 200 C
- high alumina heater plates
- graphite on the door to act as a gettering agent
- high purity forming gas at a flow rate of at least 5 volume changes an hour
- a purge with the furnace soaking at 200 C until the dew point was -25 C

This information was used in the design of the large scale furnace, which is described in Chapter 5.

3 Design of a Debinding Apparatus

Currently, two furnaces are needed for complete post processing of 3D printed parts. The first to debind green parts, which involves binder burn off and the second to sinter and infiltrate the debound parts. In addition to the additional capital investment the second furnace entails, debinding the green part in a separate furnace is more time consuming since it involves an additional ramp up and cool down.

This chapter describes the design, fabrication and testing of a debinding apparatus for use in the small scale furnace. The developed design is easily scalable and thus can also be used in the large scale furnace.



Figure 3.1: Schematic showing the concept of a debinding chamber. Note that the annular clearance between the effluent tube and the hole in the Graphite lid, t = 1/2(D-d).

3.1 Current practice

A 3D printed green part consists of powder that is held together by binder, which is typically a polymer such as Acrysol. This polymer is burnt off at around 500 C at which temperature the part undergoes light sintering, giving the part structural integrity. Binder burnout releases hydrocarbons that tend to deposit on the walls of the furnace. This makes the furnace unusable for sintering or infiltration, both of which require a contaminant free environment.

To get around this problem, the current practice is to use two separate furnaces, one for debinding and the other for sintering and infiltration. This involves an additional capital expenditure on a second furnace and increases the post processing time. Also, it requires that the debound part (that is not structurally strong) be physically transferred from one furnace to the other. Clearly, it would be ideal to have debinding and sintering take place in a single run in the



Figure 3.2: Schematic showing gas flow into the furnace and two possible exit paths for outgoing gas: one through the annular gap and out through the effluent tube and the other through the regular outlet O1.

same furnace.

3.2 Design of a debinding apparatus

In order to prevent contamination of the furnace due to binder burnout, the green part was housed in a graphite crucible with a graphite lid. The lid had a through hole in its center and was screwed on to the crucible by means of two SS screws (Figure 3.1). It should be noted that SS is not the best material to be used for the screws, since it reacts with the graphite to form cast iron. For future applications, use of Alumina screws is recommended. A SS tube passing through this hole carried the effluent generated inside the crucible out of the furnace. In order to keep the effluent generated by binder burnout from leaking out into the furnace through the annular gap between the tube and the hole, the pressure outside the crucible must be maintained higher than the pressure inside it. Forming gas entering the furnace through the bottom can get out of the furnace through the regular outlet O1, whose opening is controlled by a needle valve or through the annular gap into the graphite crucible and out of the SS effluent tube (Figure 3.2). The fraction of inlet gas going out of the furnace through the regular outlet and the effluent tube depends on the resistance the two openings offer to its flow. Let p1 denote the pressure inside the graphite crucible and p2 that in the furnace (outside the crucible). The effluent tube vents out into the atmosphere and is roughly 2m long. The gas flowing though it cools down to room temperature very soon after it leaves the furnace and can thus be assumed to be at room temperature for purposes of calculation. Pressure drop for flow through the effluent tube is given by

 $p1-pa = 8Q\mu L/(\pi R^4)....(1)$

where pa is atmospheric pressure, Q is the flow through the tube, μ is the viscosity of the Forming gas, L is the length of the effluent tube and R is the inside radius of the effluent tube. Equation 1 assumes fully developed flow in the tube. If all the Forming gas entering the furnace were to leave it through the effluent tube (worst case), then Q =8.7*10⁻⁶ m³/s. Also, at room temperature, $\mu = 2.2*10^{-5}$ kg m/s. Using L = 2 m and R = 0.085" in equation 1 gives a pressure drop of 0.2" of water over the length of the tube. Thus, for all practical purposes, the pressure inside the graphite crucible is atmospheric.

During furnace use, it is desired to maintain a pressure difference p2-p1 equal to 4" of water. The problem at hand was to determine the annular clearance, t, such that a reasonable amount of flow would pass through the annulus for the given pressure difference of 4" of water.

3.2.1 Determination of the annular clearance

The viscous pressure drop during flow through the annulus can be approximated by using the Poiseuille equation. For flow between two plates of width w and separated by a distance h, the flow rate, Q is given by

 $Q/w = h^3/12\mu^* (-dp/dx)....(2)$ where x is in the direction of flow.

In the case of the annulus (see Figure 3.1),	
$w = \pi^*(D+d)/2$	(2a)
H = t	(2b)
-dp/dx = (p2-p1)/L.	(2c)

Thus, in this case, using equations 2a and 2b, the Bernoulli pressure drop $(=1/2\rho v^2)$ is $1/2\rho (Q/wt)^2$ and the viscous pressure drop is $12Q\mu L/(w t^3)$. Neglecting Bernoulli pressure drop, which is found to be negligible compared to the viscous term, the drop p2-p1 is given by $p2-p1 = 12Q\mu L/(w t^3)$(3) Now, Q, the flow through the annulus is a fraction of Q', the flow into the furnace. At

temperatures above room temperature, Q' is related to Q_{rt} , the flow at room temperature, by

 $Q' = Q_{rt}\rho_s/\rho....(4)$

where ρ_s is the density of forming gas at room temperature and ρ is the density of the gas at the furnace temperature.

Also, let

Q = nQ'.....(5)

where n is the fraction of incoming gas that leaves the furnace through the annular gap.

Plugging equations (3) and (4) into (2) gives

 $\Delta p = 12\mu L(n Q_{rt}\rho_s/\rho)/(wt^3) \dots (6)$

The annular clearance t was estimated by assuming a pressure difference of 4" of water and that 30% of the flow went through the annulus (n = 0.3) at room temperature. At room temperature, $\mu = 2.2*10^{-5}$ kg m/s, $\rho = \rho_s = 1.546$ kg/m³, $Q_{rt} = 8.7*10^{-6}$ m³/s (=5 volume changes an hour for an 8" diameter by 8" height furnace). Choosing a diameter of ¹/₄" for the SS tube fixes w (see equation 2a). L, the thickness of the crucible's graphite cover is chosen to be ³/₄". Plugging these in equation 6 gives t = 3.5/1000".

3.2.2 Testing the design

When it came to drilling the hole in the Graphite lid, the diameter of the SS tube was measured to be 0.242" and thus the hole was reamed to a diameter of 0.249". The first test that was done with the setup (at room temperature) involved closing outlet O1 such that all the flow entering the furnace went out through the annular gap. It was found that the pressure build up in the furnace was 2" as opposed to 12" which is predicted by equation 6 for n=1. This can be explained by the fact that the SS effluent tube is not concentric with the hole in the lid. (Figure 3.3). Clearly, as figure 3.3 shows, the effective annular gap for a non-concentric case is larger than t and thus resistance to flow is lesser, which results in a lower pressure build up for a given flow. A trial and error method was used to find the gap that would yield a pressure difference of 2" in the case when all the flow entering the furnace went out through the annular gap. It was found that using a value equal to 1.83t in equation 6 for the annular gap yielded a pressure difference of 2". Thus the effective annular gap starts off at zero where the tube touches the hole in the lie in the graphite top and reaches a maximum of 2t before going back to zero. If this value of 1.83t

were correct, then the theoretically predicted and experimentally observed pressure in the furnace would agree at 500 C. To test this, when the furnace was at 500C, outlet O1 was closed and the pressure inside the furnace was noted to be 10" of water. At 500 C, the theoretical pressure difference assuming that all the flow goes out of the furnace through the annular gap (= 1.83t) can be predicted using equation 6. At 500 C, $\mu = 4.63 \times 10^{-5}$ kg m/s, $\rho = 0.58$ kg/m³ and the annular gap shrinks a little since the thermal expansion of SS is greater than that of Graphite. ($\alpha_{graphite} = 6 \times 10^{-6}$ and $\alpha_{steel} = 12 \times 10^{-6}$). Plugging these values in equation 6 yields a pressure difference of 11.3" of water that is close to the observed value of 10" of water. Thus, the assumption that the effective annular gap is 1.83t is a good one.



Figure 3.3: Schematic showing an increase in the effective annular gap between the tube and the hole due to misalignment.

It is important to note that the annular gap will vary every time the debinding mechanism is assembled before a furnace run. The annular gap could be t all around (perfectly concentric case) or could vary from zero to 2t as shown in figure 3.3 (perfectly non-concentric case) or could be somewhere between the two extreme cases. Irrespective of what the alignment is like, the following scheme can be used during any furnace run. At room temperature, Outlet O1 can be opened enough to get a furnace pressure of around 2" of water and to have enough flow through O1 to take dew point measurements. As the furnace ramps up, the pressure inside the furnace would rise, and O1 would need to be opened as required to keep the furnace pressure at 4" of water. Equation 6 can be used to see that at 500 C, when outlet O1 has been opened enough to get the furnace pressure to 4" of water, around 35% of the gas flowing into the furnace leaves via the annular gap. At 1200 C, using, $\mu = 7*10^{-5}$ kg m/s, $\rho = 0.324$ kg/m³ in equation 6 it is seen that around 13% of the flow leaves the furnace via the annular gap when O1 has been opened enough to keep the furnace pressure at 4" of water.

The next step was to debind and sinter a green part using the set up. In order to ensure that the mechanism was working and no binder from the green part leaked out into the furnace, a U tube was mounted inside the furnace (Figure 3.2). This tube carried a continuous flow of water, and thus served as a cold finger. If any polymer leaked out into the furnace, it would condense on the finger and would thus be visible after the test.



Figure 3.4: Picture of a SS truss that was successfully debound and sintered using the described set up.

A green part was placed in the graphite crucible and the cover was tightened onto the crucible using two SS screws. Boron nitride paint was smeared on to the part of the SS effluent tube that was inside the Graphite lid in order to prevent any reaction between the SS tube and the Graphite. Before the flow of cooling water was started, the furnace was purged for half an hour with forming gas at 10 volume changes an hour. Once the dew point in the furnace was below the temperature of the cooling water, cooling water flow was started. Had the flow been started earlier, water vapor inside the furnace would have condensed on the cooling tube. The furnace was then ramped up and outlet O1 was opened according to the previously described scheme.

Debinding was carried out at 500 C for one hour after which the furnace was ramped to 1200 C. Upon completion of the run, the U tube was examined and no condensate was found on the tube. This showed that there was no leakage of effluent gas out of the crucible. Figure 3.4 shows a part that was sintered and debound successfully using this setup.



Figure 3.5: Simplified schematic of a modified flow meter. Such a flow meter, when mounted at outlet O1 of the furnace can be used to maintain constant pressure inside the furnace during the entire furnace cycle without user intervention.

3.3 Improved design

The design described in the previous section works well. However, as the furnace heats up, furnace pressure begins to rise and the needle valve at outlet O1 needs to be opened in order to keep the furnace pressure at 4" of water. This needs to be done frequently as the furnace heats up, and, needless to say, is inconvenient. Figure 3.5 shows a simplified schematic of an alternative to using a needle valve at outlet O1. The idea is to use a variable area flow meter instead. The flow meter has a varying cross section passage (slightly tapered) and a float that sits over an orifice that serves as the gas inlet. The float stays in equilibrium under the influence of its own weight and the force of the gas underneath. The idea is to find the weight of the float (by trial and error) that will cause it to not rise above the orifice until the pressure of the gas is 4" of

water or more. The float weight can be altered by changing the material of the float. If this flow meter is fitted at outlet O1 of the furnace, then, as long as the pressure inside the furnace is below 4" of water, all the forming gas entering the furnace will leave it through the effluent outlet. However, as soon as the furnace pressure starts to rise above 4" of water as a result of furnace ramp up, the float will rise, causing part of the gas to exit the furnace out of outlet O1. This method will ensure that the pressure inside the furnace stays below 4" of water during the entire furnace run without user intervention.

3.4 Summary

This chapter described the design and successful application of a debinding apparatus that can be used to debind and sinter a green part in a single run without contaminating the furnace. An improved design that involves using a modified flow meter to maintain constant pressure inside the furnace without user intervention is also presented. The debinding apparatus presented in this chapter can easily be scaled up for use in the large scale furnace.

4 Directional Solidification

Interconnected porosity in 3D printed parts causes leakage in injection molding tooling and thus needs to be eliminated. It is believed that porosity is caused due to solidification shrinkage, trapped gas or evolution of dissolved gas when the liquid melt solidifies. Having optimized the small scale furnace to deliver oxide free sintered parts, the next step was to conduct experiments using this furnace to study the issue of porosity.

4.1. Porosity

4.1.1. Problems associated with porosity

The infiltrant has a lower melting point than the matrix material and gets wicked up the sintered part by capillary action. For successful infiltration, good wetting must occur between the infiltrant and the sintered skeleton. After infiltration, the resulting structure should be fully dense: however, this is almost never the case. Usually, there is porosity in infiltrated parts. Pores are undesirable since they act as crack nucleation sites, thus making the part more susceptible to failure. Interconnected porosity is worse, since it causes cooling channels in 3D printed injection molds to leak, rendering the molds unusable.

4.1.2. Causes of porosity

It is believed that the following mechanisms contribute to porosity:

- If a region in the sintered part is oxidized or contaminated, infiltrant will not wet it and thus there will be residual porosity.
- Porosity can also arise if a gas pocket is enveloped by the infiltrant melt inside the part.
- Metals shrink on solidification and this may lead to the formation of shrinkage voids. For instance, bronze, which is used to infiltrate SS skeletons shrinks 4 to 5 volume % on

solidification. If liquid is not available to feed this void, it remains unfilled and becomes a pore.

• Hydrogen is soluble in molten copper alloys. However, its solubility in solid metal is far less. Thus, on cooling, hydrogen is released, and it gets trapped, forming a pore.

4.1.3. Experiments to study and minimize porosity



Figure 4.1: Schematic showing important dimensions of the test part.

Ensuring that the sintered part has no contaminants and is not oxidized eliminates the issue of porosity due to infiltrant not reaching a given region in the part. Argon has very low solubility in molten copper alloys. Thus, switching from Forming gas to Argon near the solidus of the alloy will eliminate porosity caused by the evolution of dissolved hydrogen from the melt on cooling. This leaves the issues of enveloped gas and shrinkage porosity. Of these two it is thought that shrinkage porosity is more predominant. A series of experiments was conducted in the small scale furnace to study this issue.

4.2. Design of experiment

4.2.1. Hypothesis

Figure 4.1 shows a schematic of a part being infiltrated. When the part cools, it undergoes considerable shrinkage (~ 4 to 5 volume % in SS- Bronze systems). This shrinkage manifests itself in the form of voids. If it were somehow possible to supply additional liquid to the shrinking regions, cavities would not result. In order to achieve this, it was decided to induce a temperature gradient in the cooling part. The idea was to keep the bottom hotter than the top. Then, the top would solidify before the bottom. Any shrinkage occurring in the top would be filled by liquid metal from the bottom melt. This would considerably reduce, if not eliminate shrinkage cavities.



Figure 4.2: Illustration of the split mold concept: pre sintered metal chunks on the outside hold the two halves of the mold together.

4.2.2. Experimental set up

A split mold was made by sawing an Alumina tube into two along its length. This was done in order to facilitate easy removal of the sintered part from the mold. The two halves of the mold were held together by already sintered chunks of metal on the outside. Figure 4.2 shows a schematic of the top view of the split mold filled with loose powder and surrounded by chunks of
sintered metal. The diameter of the part being sintered was limited by the size of the small scale furnace. Before being put inside the furnace, the mold was tapped to obtain good packing of the powder. After sintering, sixteen holes were drilled into the part (Figure 4.3b). The idea was to have as many holes as possible, so that it would be easier to detect interconnected porosity (by pressurizing the holes one by one and looking for leaks through the others, when the part was immersed in water). The depth to which the holes could be drilled was limited by the fact that capillary rise of the infiltrant into the holes was not desired.

In figure 4.1, let h denote the vertical distance from the free surface of the infiltrant melt to the bottom of the drilled hole, and r, the radius of the drilled hole. In order that the infiltrant does not rise up into the holes,

 $h \ge 2\sigma \cos\theta / (\rho rg)....(1)$

where θ is the wetting angle. For the SS-Bronze and Molybdenum-Copper systems with good wetting, $\theta \sim 0$ is a good approximation. Using $\sigma = 1$ N/m, r = 0.11 inch, and $\rho = 8.99$ g/cc gives h ~ 0.315". Next, the height h1 of the free surface of the melt from the bottom of the sintered part was calculated. Refer to figure 4.1 for notation. Define the following:

M_s: Mass of sintered part

M_I : Mass of infiltrant required

 ρ_s : Density of matrix material

 ρ_I : Density of infiltrant

V_s: Volume of the sintered part: calculated by dividing the part's mass by its density

V_I: Volume of infiltrant

Then,

 $V_s = M_s / \rho_s$

 $V_I = 0.4 V_S / 0.6 * safety factor....(2)$

 M_S is obtained by weighing the sintered part. The 0.4/0.6 factor comes from the fact that the sintered part is only 60% dense. The safety factor denotes the amount of excess infiltrant that goes into the crucible with the sintered part and is typically 1.25 to 1.50. Assuming a factor of 1.50 in equation 2 gives

For d = 2", D =3.5" and V_I = 4.93 in³, h1 comes to 0.64". Using the previously calculated value of h=0.315" from equation 1, h+h1 comes to 0.955", which shows that drilling a hole that ends 1" above the melt will ensure that the melt does not rise into the holes. Thus 2" deep holes was drilled into a 3" tall sintered part (see Figure 4.3a).

4.2.3. Determination of cooling gas flow rate

In order to induce a temperature gradient in a part cooling inside the small scale furnace, it was decided to set up a jet of Argon to strike the top of the part. The jet would be started at the end of the infiltration cycle, when the furnace was ready to begin cooling down. Figure 4.3a shows a schematic of the setup. It was decided that the end of the tube carrying the Argon would be at a distance of ¹/₂" from the top of the part. In order to compute the mass flow rate of Argon that would be needed to induce a certain temperature gradient in the part, a one dimensional heat transfer model was used. It was assumed that the part lost a major part of its heat only in the x direction (see Figure 4.4). This is a reasonable assumption when the furnace heaters are on, since heat loss from the part in the radial direction is minimal.



Figure 4.3: (a) Schematic showing dimensions chosen for the test part (left) and, (b) picture of the test part (right).

Define the following notation:

MAr: Mass flow rate of Argon

Cp_{Ar}: Specific heat capacity of Argon at 1200 C

 ΔT : Temperature rise of the Argon gas from furnace inlet to outlet

k: Thermal conductivity of the part

A: Area of cross section of the part

If the temperature gradient induced inside the part in the x direction is taken to be dT/dx, then, energy balance in the x direction gives:

 $k A dT/dx = M_{Ar} Cp_{Ar} \Delta T....(5)$





This equation assumes that the gas heats up from room temperature to the furnace temperature only when it contacts the top of the part that is being infiltrated. In reality, though, the gas starts heating up from room temperature right from the moment it enters the furnace. Thus, by the time it strikes the part, it is already much hotter than room temperature. Also, equation 5 assumes that the gas heats up to 1200 C when it comes in contact with the part. This is not necessarily true. Hence, equation 5 is only approximate. However, it can be used to get an estimate of the mass flow rate of Argon needed to induce a desired temperature gradient in the part. A heat transfer analysis is presented later in this section to get an idea of the maximum amount of heat that can be transferred from the part to the gas. The material system used for initial testing had Molybdenum as the matrix material and Copper as the infiltrant. The sintered part had a height H of 3" and a diameter d of 2" (Figure 4.3a). Ignoring the area of cross section of the holes drilled out, the area A is $\pi/4d^2$, which comes to $2.03*10^{-3}$ m². The thermal conductivity of Molybdenum is around 90 W/mK at 1200 C. The specific heat of Argon, Cp_{Ar} is 520 J/kg K. The rise in temperature of the Argon can be assumed to be around 1180 C (1200 minus room temperature,

which is 20 C). Assuming that a temperature gradient of 10 C/cm is desired in the part, equation 5 gives, $M_{Ar} = 2.93*10^{-4}$ kg/s. At room temperature, the density of Argon is ρ_{Ar} 1.622 kg/m³. Thus, the flow rate needed at room temperature is $M_{Ar}/\rho_{Ar} = 23$ standard cubic feet per hour (scfh). As the Argon gets heated up, its density goes down to 0.324 kg/m³ and thus the flow rate increases. A calculation was thought to be necessary to compare the velocity of the Argon issuing out of the tube to the speed of sound at 1200 C. For a ¹/₄" diameter tube with 0.035" thick walls, the area of cross section is 1.64* 10⁻⁵ m². A flow of 23 cfh at 1200 C would be 23*1.622/0.324 = 115 cfh. The velocity at the tube outlet would thus be 55 m/s. The speed of sound in Argon at 1200 C is given by

For Argon, substituting $\gamma = 1.66$, R = 208 J/kgK gives V_{sound} = 713 m/s. Thus the velocity of the issuing jet is far below sonic and a flow rate of 23 scfh was chosen for the cooling gas.

A simple heat transfer model was used to check if the heat loss from the part to the Argon jet, given by $M_{Ar}\,Cp_{Ar}\,\Delta T$ in equation 5 was a reasonable value. For purposes of heat transfer, the part being infiltrated was taken as a flat plate, with Argon flowing over it. The length of the plate was taken to be equal to the diameter of the part being infiltrated (L~50 mm). The velocity of the Argon flowing over the plate was taken to be equal to the velocity of the Argon jet striking the part on its top (v = 55 m/s). For purposes of calculation, properties of Argon were taken at 600 C, which is roughly the average of room temperature and 1200 C, the furnace temperature. At 600 C, $\mu = 5*10^{-5}$ kg m/s and $\rho = 0.515$ kg/m³ for Argon. The Reynolds number for this flow is Re = $\rho v L/\mu = 28321$, implying that the flow is laminar. The Prandtl number of Argon at 600 C is 0.661(from gas tables). For Laminar flow, the Nusselt number $Nu = 0.664 \text{ Re}^{0.5} \text{Pr}^{0.33}$, which comes to 97.3. But, by definition, $Nu = hL/k_{Ar}$, where $k_{Ar} = conductivity$ of Argon at 600 C =0.047 W/mK. Thus, using Nu = 97.3 gives h, the heat transfer coefficient = 91.5 W/m²K. The heat transferred from the part to the gas by convection is given by $Q = hA(T_{plate}-T_{gas})$. Using A = the area of the top of the part, which is circular with a diameter of 2", $T_{plate} = 1200$ C and $T_{gas} =$ (1200+25)/2 = 612.5 C, gives $Q_{\text{convection}} = 105.4$ W. In equation 5, it was assumed that an amount of heat equal to $M_{Ar} Cp_{Ar} \Delta T = 175.2$ W would be transferred to the gas from the part. Given that Q_{convection}= 105.4 W, 175.2 W seems to be an overestimate.

4.3. Directional solidification experiments

4.3.1. Initial efforts

The first experiments were conducted on small Molybdenum parts infiltrated with Copper. These parts were 1.4" in diameter, 2" tall, and had nine 1" deep blind holes drilled into them (note that since these parts had a smaller area than that assumed while calculating the flow rate of Argon in equation 5, a temperature gradient higher than 10 C/cm is expected). When these parts were infiltrated without any cooling gas, leaks were found. Then, in the next run, cooling gas was turned on at the end of the infiltration at 1200 C, when the internal heaters were turned off. This flow was continued until the melt was solid, which occurred when the furnace temperature was around 870 C. The part was then leak tested by pressurizing the holes one by one and testing for leaks through the other holes. It was found that none of the holes leaked. Inspired by this success, the part dimensions were increased to those shown in figure 4.3 for the next run. It was found that these parts did not leak either. In order to confirm that the absence of leakage was indeed due to the cooling gas, the next run was carried out without any cooling gas. Surprisingly, it was found that the parts still did not leak. This was unexpected. A little thought into the issue showed that there was a temperature gradient within the furnace, with the temperature dropping off towards the top in the zone where the part was present. Thus, it was hypothesized that although this effect was not noticeable for small parts, since they were effectively in a uniform temperature zone, the taller parts were experiencing directional solidification, even without the use of cooling gas.

4.3.2. Reverse directional solidification

In order to examine the effect of introducing a temperature gradient in a cooling part on porosity, it was decided to compare porosity in two cases, one where the cooling gas hit the part at the top and the other where the cooling gas hit the liquid melt from the bottom. The first case is directional solidification (ds), in which the top is cooler than the bottom, and the second case is reverse directional solidification (rds), where the bottom is cooler than the top. Since the bottom cools before the top in rds, shrinkage cavities in the top cannot be fed with liquid melt and thus we expect much greater porosity in the part than in the ds case. In order to ensure that the majority of heat loss is in the x direction (as opposed to the radial direction), cooling was done at 2-degrees/min from 1200 C to 800 C (solidus of 90-10 Cu-Sn bronze is 870 C). Also, both these runs were done with a SS-Bronze system using only 25% excess melt as opposed to the 50% used in the initial trials with the Molybdenum parts. Cutting the excess melt from 50% to 25% reduced its thermal mass and made rds more effective, thus improving the contrast between porosity in parts cooled using ds and rds.



Figure 4.5: Pictures a, b and c are from the cross section of a directionally solidified part. Pictured d, e and f are from a part that was solidified using reverse directional solidification. The particles are SS, and the filler is Bronze. Pores are seen as black spots. Figure 4.5 shows cross-sections at the same height from the bottom of parts cooled using ds and rds respectively. It is evident that porosity in the rds case in much greater than that in the ds case. It was also noted that the porosity in the part cooled by ds decreased as one went from the top of the part to the bottom. This is consistent with the theory that porosity is caused by shrinkage cavities: the farther a section is from the melt, the less likely it is for liquid melt from the bottom to fill up a shrinkage cavity there.

4.3.3. Modeling surface porosity

Often, porosity occurs at the surface of a part. An attempt was made to investigate if this could be caused by shrinkage cavities within the part. Figure 4.6 shows a schematic of what powder at the surface of the part may look like. Let us assume that a shrinkage cavity forms at the location shown in the figure. Then, liquid metal from the melt rushes up through the matrix to fill the cavity. This flow up the matrix causes a pressure drop. If the pressure difference is large enough, a vacuum could be created inside the cavity. This would cause the pressure difference across the surface infiltrant film to be one atmosphere. If the film cannot sustain this pressure, it could rupture causing a surface pore. A simple way to model this situation would be to assume that flow of liquid metal to a shrinkage cavity takes place through a pipe of uniform diameter ϕ . If d_p is the diameter of the powder constituting the matrix, then, the cross section available for flow between 4 adjacent powder particles is d_p² - π d_p²/4 (Figure 4.7). The diameter d_c of a circular cross section having the same area is given by

 $\pi d_c^2 / 4 = (d_p^2 - \pi d_p^2 / 4)....(7)$

For $d_p = 30$ micron, d_c turns out to be 15.6 micron. From figure 4.1, the height of the part above the melt is (H-h1), where H is the height of the part and h1 is the distance from the bottom of the part to the top of the melt. The volume of a porous cylinder with diameter d_c extending to the top of the part is $\pi d_c^2/4$ (H-h1). Copper shrinks 5.2% on solidification. Thus the volume sucked up through the porous cylinder in order to fill the shrinkage cavity is $0.052*\pi d_c^2/4$ (H-h1). Copper freezes at 1080 C. During an infiltration run (with the Moly.-Copper system), it was observed that the liquid melt solidified when the furnace was at 1015 C, time t = 3.5 minutes after the top froze, giving the liquid melt time t to fill up any shrinkage cavities. Thus, the average flow rate of the metal that fills up the shrinkage cavity is $Q = 0.052*\pi d_c^2/4$ (H-h1)/t.



Figure 4.6: Schematic showing a shrinkage cavity formed near the surface of a cooling part. If the pressure difference across the infiltrant film at the surface is too high, the film might rupture giving rise to surface porosity.

The pressure drop due to this flow rate can be found from the equation for flow through a pipe. It



Figure 4.7: Approximation showing the cross section of a channel that the liquid infiltrant could flow through to reach a shrinkage cavity and fill it.

is given by:

 $\Delta p = 128 Q \mu L / \pi D^4 \dots (8)$

where L is the length of the channel and D is the channel diameter. Substituting L = H-h1, D = d_c and Q = $0.052*\pi d_c^2/4$ (H-h1)/t in equation 8 gives,

 $\Delta p = 0.416 \,\mu \,(\text{H-h1})^2 / (t \, d_c^2 / 4).....(9)$

where μ is the viscosity of liquid copper =3.24*10⁻³ Pa sec between 1100 to 1200 C. Using H-h1 = 2.36", $\mu = 3.25*10^{-3}$ Pa sec, and an observed value of t = 3.5 minutes in equation 8 gives $\Delta p = 410 \text{ N/m}^2$. The pressure difference across the surface infiltrant is maintained by surface tension. If good wetting is assumed between the powder and the infiltrant and R is the radius of the power particles, then the pressure difference that can be supported across the interface is given by:

 $\Delta p = 2 \sigma/R....(10)$

In the equations above, σ is the interfacial surface tension = 1.35 N/m and R is the radius of the spherical powder particles = 15 microns. Substituting these in equation 10 gives $\Delta p = 1.8*10^5$ N/m². Clearly, a pressure difference of 410 N/m² caused by flow of the liquid melt can easily be supported by the infiltrant. In fact, there is plenty of room for safety since there is a three orders of magnitude difference between the two values. Thus, even if the liquid got sucked up at 100 times the rate assumed in equation 9, the pressure difference caused across the film would be 0.41 atmospheres, around 4.5 times less than the 1.8 atmospheres that can be supported. A much more severe case would be the creation of a vacuum due to shrinkage right at the surface of the part. This would cause a pressure difference of 1 atmosphere across the infiltrant. Clearly, even this can be supported across the infiltrant without rupture. Thus, surface porosity is not caused due to rupture of the infiltrant film at the surface of the part.

4.3.4. Temperature profiles

In order to be certain that the flow of cooling gas was indeed causing the predicted temperature gradient in the part, the temperature inside the part was recorded as a function of the depth from the top. To do this, two 2" deep holes were drilled into a sintered SS part (Figure 4.8). The x coordinate was measured from the bottom of the hole (x = 0). Measurements were made for x = 0.25" to x = 1.5". Temperature was not measured at the very top of the hole since it

would not be a true representation of the temperature of the part (due to convection heat loss from the thermocouple to the cooling gas).



Figure 4.8: Schematic showing the two holes that were drilled into a part to do temperature profiling inside the part.

Temperatures were recorded using a bare wire thermocouple that was sheathed in a dual hole Mullite sheath. The thermocouples were of type K and had a wire diameter of 0.02". The Mullite sheath had an outer diameter of 1/8". Figure 4.9 shows a cross section of the sheath. The Mullite sheathed thermocouples were let into the furnace through fittings on the furnace cover. The fittings were Aluminum pieces with a slight clearance hole for the Mullite sheaths to pass through. Friction between the sheaths and the holes was enough to keep the sheaths at a given height and prevent them from sliding down. Figure 4.10 shows a picture of the setup used.





The first profile was obtained when the furnace was at 1200 C and in steady state with no cooling gas. Figure 4.11 shows the variation of temperature in the part with height. A gradient of 2.2 C/cm exists in the part even when the cooling gas is not cooling the part. The other profile in the figure was obtained after cooling gas was turned on and steady state was reached at 1200 C. In order to avoid errors due to differences among thermocouples, the same thermocouple was used to get both the profiles shown in the figure.



Figure 4.10: Picture of the set up used for temperature profiling. Note the Mullite sheath and the Aluminum clamps.

It can be seen that the gradient in the part increases to 7.24 C/cm when the cooling gas is turned on. Thus, the cooling gas has induced an additional gradient of 7.24-2.2 = 5.04 C/cm. This is just 50.4% of the theoretically estimated value of 10 C/cm that should have been induced inside the part. This can be explained by looking at the heat transfer analysis in section 4.2.3. While calculating the mass of Argon required, it was assumed that $M_{Ar} Cp_{Ar} \Delta T = 175.2$ W was removed from the part by the cooling gas. However, the heat transfer calculation in section 4.2.3 shows that convection can remove just 105.4 W form the part. If this were true, then the gradient would be 105.4/175.2*10 C/cm = 6 C/cm, which is close observed value of 5 C/cm. Thus, what is probably happening is that convection heat transfer is limiting the amount of heat withdrawn from the part, thus causing a lower than expected temperature gradient in the part.



Figure 4.11: Graph showing the variation of temperature inside an infiltrated part with and without cooling gas. The cooling gas induces a gradient of around 5 C/cm in the part. The 1200 C in the legend indicates the furnace thermocouple reading in steady state.

Once it was established that the cooling gas did induce a temperature gradient, profiling was done inside a cooling part. At the end of the infiltration cycle at 1200 C, the heaters inside the furnace were turned off and cooling gas was started. Since the temperature in the part needed to be measured continuously with time, two thermocouples were used. They were placed at different heights ($x = \frac{1}{4}$ " and $x = 1 \frac{1}{2}$ ") in the two holes drilled into the part. In steady state, without gas flow, the temperature recorded by the bottom thermocouple was 1228 C and that by the top one was 1214 C. The difference in temperatures between top and bottom previously recorded using the same thermocouple was 7 C. Hence, any difference in temperature gradient along the height of the part as a function of time. Also shown is the temperature recorded by the control thermocouple of the furnace. Cooling gas flow was turned on 15 seconds before t = 0. This caused the temperature gradient inside the part to begin increasing sharply from the 2.2 C/cm that existed before the cooling gas was turned on. The sharp increase in temperature gradient that can

be seen at t = 0 is a manifestation of this. The temperature gradient stabilized at around 13 C/cm around 2 minutes after gas flow was started. Around 75 minutes after turning on the gas flow, it was turned off. A steep drop in the temperature gradient inside the part is clearly seen in the figure.



Figure 4.12: Temperature profile measured in a cooling part. A positive gradient indicates that the top is cooler than the bottom. Note the effect the cooling gas flow has on the gradient. The sharply increasing temperature gradient seen at t=0 is a transient caused by turning the cooling gas on 15 sec before t=0. Also shown in the graph is the furnace temperature.

As long as the cooling gas was on, the average temperature gradient in the cooling part was around 13 C/cm, a good value.

4.4. Summary

This chapter described experiments that were conducted to study the issue of porosity that occurs in parts during infiltration. It was shown that inducing a temperature gradient in a cooling part reduces porosity. Temperature profiling was done in order to study the temperature gradient in a cooling part. A possible mechanism for the occurrence of surface porosity was investigated.

5 Design and Fabrication of the Large Scale Furnace

This chapter describes the design and various features of the large scale furnace. A lot of its features were scaled up from the small scale furnace. However, some of the problems encountered while designing the large scale furnace were not seen during the design of the small scale furnace. Solutions to these design problems were obtained by experimenting with various alternatives. This chapter describes issues relating to the thermal and mechanical design of the large scale furnace.

5.1. Thermal design

Any furnace needs power for two reasons: to ramp up to the operating temperature and to overcome steady state heat losses at that temperature. A simple approach was used to estimate the power requirement of the large scale furnace. Steady state heat loss was experimentally determined for the small scale furnace and was scaled up based on surface area to estimate steady state heat loss for the large scale furnace. The power required to ramp up was estimated by considering all the thermal mass that would go into the furnace. The following sub sections describe the process.

5.1.1. Estimating steady state heat losses

The small scale furnace has two semi cylindrical internal heaters, each rated to run off 115V AC. The resistance of each heater plate was measured to be 16 ohms. Thus, the total power of the two heaters combined is $2V^2/R=2(115)^2/16=1.8$ kW. When the small scale furnace was at 1200 C, with the external band heaters turned off, the controller output in steady state was noted to be 53.5%. Thus, the steady state heat loss was $0.535*1.8kW\sim1kW$. Figure 5.1 shows a schematic of the hot faces of the Alumina insulation that goes into the small scale furnace. Figure 5.2 is a schematic of the resistance to heat loss from the hot face of the insulation that is at 1200 C to the ambient atmosphere. Assuming that the convective heat transfer resistance for heat loss from the furnace wall to the atmosphere is independent of the furnace size, the steady state heat loss from the large scale furnace will simply scale as the area of the insulation exposed to

1200 C. The surface area of the insulation exposed to 1200 C was calculated to be around 340 square inches for the small scale furnace.



Figure 5.1: Schematic showing dimensions of the hot faces of the insulation that goes into the small scale furnace. These dimensions were used to calculate the area of the hot zone of the furnace.

The interior dimensions of the large scale furnace were estimated by using the fact that the part piston that would go into the furnace would have dimensions of 7" by 9" by 13". The maximum size of the part that can be printed into the part piston is 6" by 8" by 12". Thus, this would also be the maximum size of a part that would be post processed in the furnace. To account for temperature non uniformity near the furnace walls due to heat loss from the walls to the outside, it was decided to have a dead zone on all sides between the part and the insulation (see Figure 5.3). Adam Lorenz's [4] experience with profiling temperature in C. I. Hayes's furnace showed that a dead space of 3" was a good value to use.



Figure 5.2: Resistance to heat transfer from the hot zone that is at 1200 C to the ambient atmosphere. Heat transfer involves conduction through the insulation, conduction through the Aluminum shell and convection to the atmosphere.

Based on specifications from a heater manufacturer's catalog, it was inferred that heater plates mounted on the sidewalls would be at the most an inch thick. Thus, the width of the heated zone was estimated to be 7" (piston width) +2*3" (dead space) +2*1" (space for two heaters) = 15". The height of the heated zone was estimated to be 9" (piston height) +2*3" (dead space) =

15". While estimating the depth of the heated zone, a 3" dead space was assumed in front of the piston and a 5" space was taken behind the piston, in order to accommodate an infiltrant crucible that might be used during a gated infiltration. Part of the infiltrant crucible would be located in a non uniform temperature zone: however, this was not an issue as long as all the infiltrant melted. Thus, the depth of the heated furnace zone comes to 21". The dimensions of the interior heated space of the large scale furnace were thus estimated to be 15" by 15" by 21". The corresponding surface area of the insulation exposed to the interior heated space is thus 1710 square inches. Scaling up from the small scale furnace according to surface area, the steady state heat loss form the large scale furnace comes to 1 kW*(1710/340)=5.1 kW.



Figure 5.3: Schematic showing dead space between the part piston and the insulation. Dimensions of the insulation were estimated by assuming values for the dead space between the piston and the hot face of the insulation. The dimensions thus obtained were used to determine steady state heat losses by scaling up those from the small scale furnace based on the area of insulation exposed to 1200 C.

5.1.2. Estimating ramp up and total power requirement

The ramp up power requirement of a furnace depends on the thermal mass inside it. Thermal mass is the product of the mass of a given entity and its specific heat capacity and has units of J/K. In order to obtain a rough estimate of the thermal mass of the furnace, assumptions were made about the size of the shell, weight of the heater plates that would go into it, etc. Table 5.1 shows a computation of the thermal masses of various parts of the furnace. In this table, it is important to note that if external strip heaters are not used, the Aluminum shell only heats up to 200 C when the furnace heats to 1200 C, and thus its thermal mass is effectively 1/6th of the

theoretical value. However, since the normal practice would be to maintain the shell at 200 C using external strip heaters, the thermal mass of the shell would not really contribute to the thermal load of the internal heaters during ramp up. If the external strip heaters are on, then the Alumina insulation in contact with the shell is maintained at 200 C throughout the furnace cycle, while the insulation facing the internal heated area of the furnace ramps to 1200 C from 200 C. If a linear temperature gradient is assumed to exist inside the insulation, then the effective thermal mass of the insulation is roughly half the theoretical value. While determining the mass of the infiltrant, it was assumed that the part was 40% porous and that 30% extra infiltrant would be used. The net thermal mass of the furnace comes to around 71.5 kJ/K, including the effective thermal mass of the Aluminum shell. For a ramp rate of 5 K/min, the power required is mcdT/dt = 71.5*5/60 kW = 5.96 kW.

S. No.	Part name	Material	Dimensions	Mass	Spec. ht.	Thermal mass
					capacity	
				kg	kJ/kgK	kJ/K
1	Part piston	Graphite	7"*9"*13" outer dim.	8.84	0.709	6.30
			1/2" wall thickness			
2	Aluminum shell	Aluminum	22*18 5"*24 5" outer	86.5	0.903	13.02
	7 Hummun Shen	Addinindin	dim 3/4" thick	00.5	0.705	15.02
	Note: 13.02 k I/K is 1	/6 th of the obtaine	d value and is to be used if e	xternal heat	ers are not or	
	If external heaters are used, do not consider shell's thermal mass.					
3	Heater plates	Alumina	Estimated from sample	12.96	0.765	9.91
4	Dort for sintaring	22	6"*10"*8" 600 damaa	44.0	0.49	21.22
		00	0 12 8,00% uense	44.2	0.46	21.22
5	Infiltrant	Bronze	40% dense part and	44	0.355	15.62
			30% extra infiltrant			
6	Hearth plate	Graphite	14"*8"*1"	2.3	0.709	1.63
7	Infiltrant crucible	Graphite	3.75"*7"*9" outer dim	3.8	0.709	2.69
		onpinto	1/2" walls		01105	2.03
8	Insulation	Alumina	Inner dimensions of	1.5	0.765	0.57
No	te: see text for discuss	ion	shell, and 1" thick			
			insulation			
9	Catch tray	SS	Estimated mass	1	0.48	0.48
					TOTAL	71.45

Table 5.1: Estimation of the thermal mass of a fully loaded furnace.

Having estimated the ramp up power requirement, the next step was to calculate other heat losses from the furnace. Power would be supplied to the internal heaters using a brass tube brazed to the heater leads (Figure 5.9 and Figure 5.16). Heat is lost from the inside of the furnace to the outside by conduction along the heater lead and the brass tube. The heater lead has a lower conductivity than brass. An upper bound estimate of the heat loss can be obtained by assuming that the conductivity of the heater lead is equal to that of the brass tube. The brass tube has an outer diameter of 0.125" and a wall thickness of 0.015" which gives it an area of cross section of $A = 2.9*10^{-6} \text{ m}^2$. As a worst case analysis for estimating heat loss, it can be assumed that the temperature drop of $\Delta T = 1000 \text{ K}$ occurs entirely along a length L = 1" of the insulation. The corresponding heat loss by conduction is kA $\Delta T/L$, where k is the thermal conductivity of brass = 111 W/mK (@room temperature, worst case scenario). This heat loss comes to around 13W. For 24 feedthroughs, the total heat loss by conduction comes to 312 W. To account for heat loss through other fittings in the furnace by conduction, the total heat loss by conduction can be taken to be around 1kW.



Figure 5.4: Picture of the large scale furnace. Note the strip heater and the view ports on the front door.

Thus, the power requirement for the furnace is 5.1 kW (steady state heat loss at 1200 C) + 5.96 kW (ramp up power requirement at 5 K/min) + 1 kW (loss through power feedthroughs etc.) = 12.06 kW.

5.1.3. Selection of internal heaters

Thermcraft Inc., of Winston-Salem, NC was chosen to provide heaters for the furnace. The heaters had Fe-Cr-Al, "super Kanthal" resistance wire (able to reach 1350 C) embedded in Alumina plates and were of a similar kind as the ones used in the small scale furnace. Due to considerations of size and watt density, the heater plates chosen gave a maximum power of 850 W. Twelve such heaters were ordered, giving a total power of 10.2 kW. At first glance, this power seems lower than the required 12.06 kW. While arriving at the 12.06 kW, it was assumed that the charge inside the furnace would be the maximum possible that could be accommodated by the piston. However, this is almost never the case, and thus the ramp power requirement will be less than the calculated 5.96 kW. Also, the steady state requirement of 5.1 kW is when the furnace is at 1200 C: at lower temperatures, the power requirement to offset heat loss is lower and thus more heat is available to ramp up. Moreover, as discussed earlier, with the external strip heaters on, the thermal mass of the Aluminum shell shall not be heated by the internal heaters. This will reduce the ramp up power requirement by 1.1 kW. Thus, it was decided to stick with the twelve 850 W heaters.

5.2. Furnace shell and external features

Aluminum was retained as the shell material. This was because of its excellent thermal conductivity that would enable the shell to attain uniform temperature when heated by external strip heaters. Since the shell would attain a temperature of around 200 C when the furnace was running, it was important for the shell material to exhibit good strength at this temperature. Three potential candidates: Aluminum 5052, 5083 and 6061 were examined for suitability for use as shell material. At 200 C, the yield strengths of Aluminum 5052, 5083 and 6061 are 75 MPa, 117 MPa and 103 MPa respectively. At 260 C, these numbers become 52 MPa, 75 MPa, and 35 MPa respectively. Clearly, Aluminum 5083 is superior to the others in terms of strength. This, together with the fact that it exhibits good corrosion resistance made it an ideal candidate for shell material.

As mentioned previously, shell dimensions were chosen keeping in mind the fact that an entire part piston having dimensions of 7" by 9" by 13" would need to be fired in the furnace. Also, dead space was provided around the periphery of the part where the temperature would not be uniform due to greater heat loss at the ends. An inch of Alumina insulation was used just as in the small scale furnace. These guidelines were given to a consulting engineer, Diane Brancazio, who was staffed on this project in order for her to design a rectangular cross section shell.



Furnace shell

Figure 5.5: Picture showing a top view of the furnace shell. Note the QF40 fittings and the thermocouple feedthroughs.

A shell thickness of ³/₄" was chosen and the shell was designed using Solidworks, a CAD software. Features such as standoffs on the furnace sidewalls for power feedthroughs, flange fittings, gas ports, and mounts for internal heaters were incorporated into the shell. Design parameters for features on the shell were determined as the design progressed and parts such as heaters, thermostats, burst disc etc. chosen to go into the furnace were finalized. Figure 5.4 shows a picture of the shell. The outside dimensions of the shell are 18.5" by 22" by 24". The shell was fabricated by NU Vacuum Systems Inc., a Massachusetts based vacuum chamber manufacturing company. The shell was welded to be leak tight and was vacuum tested by the manufacturer. Detailed engineering drawings of the furnace and its subassembly are included in

Appendix 2 of this document. The sections below briefly describe the internal and external features of the furnace.

5.2.1. External strip heaters

Figure 5.5 shows a top view of the furnace shell. Two strip heaters (1 kW each) can be seen on the top face. There are two such heaters on each face of the furnace, except on the door that has only one heater, due to space considerations. Each strip heater runs off 240 V AC and is controlled by its own normally closed thermostat which opens at 200 C. The strip heaters are held against the shell by screws at the ends and a clamp in the middle (Figure 5.5). Heat transfer paste (see list of vendors in Appendix 1 for details) between the strip heaters and the Aluminum shell ensures good heat transfer between the two. High temperature wires (rated for use at 400 C, see list of vendors in Appendix 1 for details) carry current to the strip heaters and run along the furnace shell. Strain relief clips have been provided on the shell at regular intervals to relieve these wires.

5.2.2. Fittings on the shell

Also seen in Figure 5.5 are six QF40 flange fittings on the top face. The bottom face has two similar fittings, one of which is fitted with a burst disc designed to rupture at 4.5 psig. The QF40 fittings can be used for various purposes such as: a passage for a linear feeding mechanism to be used for gated infiltration, for the effluent tube of a debinding apparatus, or as ports for additional thermocouples. The figure also shows two type K thermocouples: the front one is the control thermocouple. The thermocouples go into the furnace through standard ¹/₄" NPT compression fittings. A gas outlet port can be seen in Figure 5.4 on the top face of the furnace. Four such ports have been provided, two on the top face and two on the rear face for use as gas inlet or outlet ports. Gas tubing goes into the furnace through 3/8" NPT compression fittings.

5.2.3. Furnace Doors and heat shields

Figure 5.6 shows a comparison of the two doors that can be used with the furnace. Both doors are fastened on to the shell using eight SS bolts. A provision was made at four locations on the front face of the furnace shell for mounting door clamps to clamp the door to the shell (Figure 5.4). However, it was found that the clamps interfered with the opening of the door and were thus not used. In Figure 5.6, the door on the right has two view ports and is meant for use during regular operation of the furnace. The door shown on the left has six ports for thermocouples that can be used to measure temperature profiles inside the furnace. This door is to be used whenever the temperature inside the furnace needs to be measured and adjusted along the depth of the furnace by using trim potentiometers (more on this in Chapter 7). Figure 5.7 is a picture of the furnace fitted with this door.



Figure 5.6: Comparison of the two kinds of doors that can be used with the furnace. The left door is for use during temperature profiling and the right door with the viewports is for use during regular furnace runs.

Figure 5.4 shows two G-10 heat shields, one on each side of the furnace. G-10 is a woven glass fabric laminated with epoxy resin, can operate continuously at 140 C, and is a superior electrical insulator. The purpose of the G-10 shields is to prevent radiation from the hot furnace wall (at 200 C) from reaching the surrounding furnace electrical control box and gas tanks. These

shields are held on to the furnace wall by means of steel standoffs that screw into blind threaded holes in the furnace wall.



Figure 5.7: Picture of the furnace fitted with a door used for temperature profiling. The door has six ports for inserting thermocouples into the furnace. The thermocouples can be used to measure the temperature inside the furnace as a function of the distance from the door.

5.2.4. Power connections to internal heaters

There are twelve internal heaters, six on each side wall of the furnace. These heaters need to be connected to power leads from the SCRs. The heater leads were brazed (leak tight joint) to 1/8" outer diameter Brass tubes using a Copper-Silver eutectic alloy (see Figure 5.8 and Figure 5.9). An important issue was to bring the Brass tubes out of the furnace while maintaining a leak proof seal. Conventional methods such as using a Swagelok compression fitting and a Teflon sealant did not work. This was because the Teflon crept under the influence of the high shell temperature and the force of the compression fitting. Figure 5.10 shows the Swagelok fitting-Teflon sleeve assembly that was used to test if this assembly would work at 200 C. As will be explained later in the chapter, as part of preliminary testing, a single heater was mounted into the furnace and an enclosure was created around it in order to test if the heater could heat the enclosure up to 1200 C. When this test was done, Swagelok assemblies like the ones shown in

Figure 5.10 were used to create leak proof feedthroughs. Figure 5.11 shows the resulting creep in the Teflon sleeve of one such assembly. A second series of tests involved tightening assemblies like the one shown in Figure 5.10 and placing them in an oven at 200 C for a few hours and examining the behavior of the Teflon sleeve. Creep was observed in the sleeves.



Figure 5.8: Picture showing an internal heater with its leads braze joined to Brass tubes. The heater leads stick ¹/₂" into the Brass tubes. The braze is a Silver-Copper eutectic.



Figure 5.9: Picture showing the braze joint between the heating element and the Brass tube. Once outside the heater plate, the heating element is double twisted to reduce resistive heating. This keeps the braze joint cool.

To overcome this problem, fittings shown in Figure 5.12 were machined out of Ultem. Ultem is a plastic that has excellent machinability, mechanical strength, and dimensional stability. Moreover, it is electrically insulating and can withstand temperatures up to 170 C. As can be

seen in Figure 5.12 and Figure 5.13, the fitting has ¹/₄" NPT male threads on one end, a small projection on the other, and a through hole that is slightly bigger than 1/8" in diameter (0.128"). The Brass tube slides into the fitting through this hole. A detailed engineering drawing (drawing # FUR025) of the Ultem fitting is included in Appendix 2. A Silicone rubber tube



Figure 5.10: Picture showing the Swagelok fitting and Teflon sleeve assembly that was tested as a possible candidate for use as a power feedthrough.

having an ID of 3/32" and an OD of 5/32" is slid over the Brass tube and on to the projection on the Ultem part to form a leak proof seal (Figure 5.13).



Creep in Teflon sleeve at the location where the ferrules made the seal

Figure 5.11: Picture showing creep in the Teflon sleeve that was used in a Swagelok fitting and maintained at 200 C for a few hours.

The seal formed by the Silicone tube and the Brass tube was tested for leaks under positive pressure and was found to not leak even at 15 psig. Moreover, this seal will perform very well when the furnace is run under vacuum, since atmospheric pressure outside will force the Silicone sleeve against the brass tube, resulting in a more secure seal. Figure 5.14 shows how a current



Projection with lip for mounting the Silicone

Figure 5.12: Ultem fitting which was used as a power feedthrough for heater wires.

carrying wire from an SCR is connected to the Brass tube using a Brass connector. Figure 5.15 shows a drawing of the brass connector. Note the slit in the middle. Force from a clamping screw decreases the slit spacing and clamps the connector firmly on to the brass tube. Figure 5.14 shows the how the Ultern fitting threads into an Aluminum standoff welded on to the furnace shell. There are twelve such standoffs each on the left and the right faces of the furnace shell, some of which can be seen in Figure 5.4. Figure 5.16 shows a schematic of a cross section of the standoff. The Mullite tube inside the standoff is to prevent the heater wire from accidentally coming in contact with the Aluminum shell.



Figure 5.13: Picture demonstrating how a Silicone tube can be used to achieve a leak proof joint at the Ultem fitting-Brass tube interface.

5.3. Insulation

5.3.1. Initial efforts

The small scale furnace used an inch of Alumina as insulation. Mounting the Alumina insulation into the small scale furnace was relatively easy since most of the insulation was well secured and did not have to be held against its weight. However, attempts to mount Alumina



Figure 5.14: Picture showing the power feedthrough assembly. The wire from the SCR is connected to the Brass tube by a Brass connector. The Ultem fitting is screwed into the Aluminum standoff.

insulation into the large scale furnace proved difficult. This was especially true of the insulation on the top, back and front faces of the furnace that had to be held against its own weight.



Figure 5.15: Drawing of the Brass connector used to connect a wire from the SCR to a Brass tube that is brazed to a heater lead. Holes A and B in the drawing are clearance holes for 8-32 and 6-32 screws respectively.



Figure 5.16: Schematic showing a section through an Aluminum standoff on the furnace wall. The Mullite tube inside the standoff is to prevent the heater wire from touching the Aluminum shell.

The problem was that the 1" thick Alumina blanket from which insulation was cut was actually composed of many thin sheets of Alumina that were stitched to each other by fine thread. When insulation of the desired size was cut from a roll of Alumina blanket, the fine threads that held the sheets together got cut and thus the insulation got ripped off easily. When this insulation was mounted into the furnace, it did not retain structural integrity. Insulation on the top, front and back faces sagged and peeled off. Peeling was not an issue for insulation on the sides, since the heaters held it against the Aluminum shell of the furnace. Clearly, peeling was an issue that had to be addressed.

5.3.2. Polymer spray

There were two requirements: one, the insulation had to be stiff at room temperature so that mounting would be easy and two, the insulation needed to be held in place even when the furnace was at 1200 C. In order to attain stiffness at room temperature, it was decided to spray the face of the insulation that would be exposed to the hot zone with polymer. The polymer

would disintegrate at high temperatures and the insulation would be free from the polymer after the first furnace run. The cold face of the insulation was not sprayed with polymer since it would not get hotter than 200 C, at which temperature the polymer would not disintegrate. The important thing was to search for a polymer that would disintegrate at 1200 C without leaving a trace. Three polymeric binders were tried: Rhoplex, Polyethylene Glycol (PEG) and Polyacrylic Acid (PAA).

In order to compare these binders, the hot face of the top insulation of the small furnace was sprayed with an 11 weight % binder solution and fired to 1200 C. This process was repeated with fresh insulation for all three binders. In Figure 5.17, the black coating on the insulation on the left was obtained when it was sprayed with Rhoplex and ramped up to 1200 C at 5 C/min. This might have occurred since the Rhoplex did not have sufficient time to decompose at lower temperatures. In subsequent runs, the furnace run involved a 90-minute soak at 500 C before ramping to 1200 C at 5 C/min and staying there for 60 min. The idea was to give the binder sufficient time to decompose at 500 C before ramping up to 1200 C. The insulation on the right in Figure 5.17 shows the result of this treatment on Rhoplex sprayed insulation. As can be seen, there is binder residue on the insulation and this is unacceptable. When a similar experiment was carried out using PEG, there was no residue on the hot side of the insulation. However, polymer vapors did condense on the cold face of the Aluminum top door. This too was unacceptable. PAA on the other hand, worked well and left no visible trace in the furnace. Thus it was decided to spray the insulation that went into the large scale furnace with an 11 weight percent solution of PAA (See list of vendors in Appendix 1 for details of the PAA solution).

5.3.3. Nextel thread and Inconel mesh

The PAA coating on the insulation took care of structural integrity issues at room temperature and made mounting the insulation into the furnace an easy task. However, as mentioned earlier, the insulation also needs to retain structural integrity at high temperatures, after polymer burn off. 3M manufactures a product called Nextel, which is a high temperature ceramic capable of continuous operation at 1371 C (see list of vendors in Appendix 1 for details). The two most common grades of Nextel are types 312 and 440. Nextel 440 was chosen

over Nextel 312 for use in the furnace since it has a higher Alumina content (70% vs. 62%) and because its filament tensile strength is higher (2000 MPa vs. 1700 MPa). Nextel 440 thread and



Polymer coated and fired insulation

Figure 5.17: Picture of the results of firing insulation sprayed with Rhoplex to 1200 C. The insulation on the left was ramped to 1200 C at 5 C/min starting at room temp., while that on the right was held in the furnace at 500 C for an hour before ramping to 1200 C at 5 C/min. The idea was to give time for complete polymer burnout at 500 C before ramping to 1200 C. Both these results were unacceptable.

cloth are available in two varieties: with sizing and without sizing. Sizing is a chemical coating that makes the Nextel thread easy to handle while sewing. The sizing burns off at around 300 C. However, sized thread could not be used to sew the insulation since parts of it away from the hot zone of the furnace (i.e. facing the Aluminum shell) would not get hot enough to burn off the sizing. Hence, it was decided to use unsized thread to sew the periphery of the insulation in order to prevent it from delaminating (Figure 5.18). Unsized Nextel sleeve was sown on the insulation on the front face of the furnace. This was done to protect it from tearing apart while loading and unloading parts in and out of the furnace. In addition to sewing the insulation, Inconel wire mesh was used on the back, front and top faces to hold the insulation in place (Figure 5.19). Refer to the list of vendors in Appendix 1 for details on the mesh.

5.3.4. Initial testing

As seen in section 5.1, the installed capacity of the internal heaters was a little less than the theoretical power estimate (10.2 kW vs. 12.06 kW). Hence, it was decided to conduct experiments to determine if extra insulation would be required in order for the furnace to reach



Figure 5.18: Picture showing Alumina insulation sewn together along the periphery using Nextel thread.

1200 C. When all twelve heaters are mounted into the furnace, the area heated by them is composed of the top, bottom, front and back faces of the furnace insulation. The heated zone has dimensions of 13" by 19" by 20", and thus the area heated by the twelve heaters comes to 2(13*19+13*20)=1014 square inches. The area per heater is thus 1014/12=84.5 square inches. A single heater was mounted into the lower left corner of the furnace and a box was built around it using 1" Alumina insulation, such that the area heated by the heater would be close to 84.5 square inches. Figure 5.20 shows a schematic of the enclosure.

A simple calculation shows that the area of the insulation exposed to the heater is 99 square inches. The heater was run off a variac and it was found that 133 V needed to be applied across the heater in order to attain 1200 C inside the enclosure. This was $(133^2/115^2-1)*100 \sim 34\%$ extra power over the rated 850 W. The extra power requirement could have been due to the fact that the heater was heating 99 square inches as opposed to the 84.5 square inches it would heat when all twelve heaters were running. Also, there were gaps at the interfaces between the insulation through which heat loss could have been occurring. However, to be on the safe side, it was decided that extra insulation (in addition to the 1" of Alumina already in the furnace) would need to be added to the furnace to make sure that it would reach 1200 C when all twelve heaters were installed.



Figure 5.19: Interior of furnace. Inconel wire mesh supports the insulation on the top, back, and front faces of the furnace. Nextel sleeve on the periphery of the front insulation protects it from being damaged when the furnace is being loaded/unloaded. Also note how the heaters are mounted on the Inconel supports.



Figure 5.20: Schematic showing a section through the enclosure that was built around a single heater. This heater was powered by a variac and the voltage required to attain a steady state temperature of 1200 C inside the enclosure was found as a function of the quantity of insulation on the 9" by 7" face opposite the heater plate.

5.3.5. Choice of extra insulation material

Choice of material for the extra insulation was critical since adding extra insulation meant adding extra moisture that needed to be purged before a furnace run. Two candidates were considered: Alumina and Graphite. Fiber Materials Inc. of Biddeford, ME offers two kinds of Graphite insulation to choose from: CH grade Carbon felt that has a thermal conductivity of 0.053 W/mK (at 50 C) and GH grade Graphite felt with a thermal conductivity of 0.068 W/mK (at 50 C). Although Carbon felt is better than Graphite felt in terms of thermal conductivity, it has a lower percentage of Carbon (94% vs. 99.7%) and absorbs a higher amount of water vapor when exposed to the atmosphere (5% vs. 2% of weight) than the Graphite felt. Thus, GH grade Graphite felt was chosen over CH grade Carbon felt. The next step was to compare the Graphite felt with Alumina. In order to compare their insulating properties, first, a second layer of 1" Alumina was added behind the already existing 9" by 7" Alumina insulation on the face of the enclosure around a single heater (described in section 5.3.4). It was found that 115 V was required to attain an enclosure temperature of 1200 C. Next, an inch of Graphite was added instead of the Alumina, and it was found that 115 V supplied to the heaters resulted in an enclosure temperate of only 1086 C. This data pointed to the fact that, at 1200 C, Alumina has better insulating properties than Graphite.

Another important criterion was the moisture absorptivity of Alumina vs. Graphite. In order to compare the two, the idea was to heat chunks of each type of insulation in a furnace and observe moisture release by monitoring the dew point inside the furnace. A tube furnace, consisting of a Mullite tube and SS heat shields at both ends was used to fire the insulation. SS heat shields were used instead of the standard ceramic heat shields in order to avoid corruption of test data due to moisture release from the ceramic shields. Also, an Argon atmosphere was used in these tests in order to prevent any chemical reaction between the insulation and the process gas.



Figure 5.21: Dew point inside the furnace vs. time for a sample of Alumina insulation fired in the tube furnace. Note the humps in dew point when the furnace temp. is 400 and 600 C.

First, the empty tube furnace was heated at 5 C/min to 1200 C and cooled down. Then, the furnace was opened for exactly 5 minutes and a sample of Alumina insulation was placed in it. This 5-minute opening time was used whenever the furnace was opened between runs in order to ensure that the same amount of moisture diffused into the furnace from the atmosphere each time it was opened. Also, all the runs were done in a 24-hour period to minimize the effect of changes in ambient humidity conditions on the outcome of the runs. The furnace was then ramped up at 5 C/min and held for 15 minutes each at 300 C, 500 C, 700 C and 900 C before ramping down to room temperature. The hold at various temperatures was done to ensure that all the moisture that could be expelled at the given temperature was expelled before the ramp continued. Figure 5.21 shows the dew point as a function of time with Alumina inside the furnace. After this run, the furnace was again opened for 5 minutes and the Alumina was taken out. Next, as a control run, the empty furnace. The next run was done with the same volume of Graphite insulation as Alumina used in the first run. Figure 5.23 shows the dew point curve for the run with Graphite.

It can be seen from figure 5.22 that for the run with the empty furnace, the dew point keeps falling with time. However, in the run with the Alumina insulation (figure 5.21), humps corresponding to moisture release can be noticed, with peaks in dew point at furnace temperatures of around 400 C and 600 C. In the case of Graphite (figure 5.23), no such humps are seen. Moreover, the average dew point level is much lower than in the case with the Alumina insulation. These tests established that Graphite absorbed less moisture than Alumina.



Figure 5.22: Dew pt. vs. time for the empty furnace. This is a control run. Note the absence of humps in dew point unlike in Figure 5.21.

It was decided to use Graphite for the extra insulation despite the fact that Graphite was a poorer insulator than Alumina. This was because of the fact that it absorbed lesser moisture than Alumina when exposed to the atmosphere. Before Graphite was used, though, another issue needed to be investigated. The heaters used in the furnace need to be fired in air at 1200 C for a few hours the first time they are used. Thus, if Graphite insulation were to be used behind the (already existing) 1" of Alumina insulation, then oxidation could be an issue. The plan was to use ¼" of Graphite insulation behind the 1" of Alumina insulation. Thus, if the furnace were at 1200 C and the shell were at 200 C, assuming roughly similar conductivities for Graphite and Alumina, the temperature of the Graphite-Alumina interface would be 400 C. A chunk of

Graphite insulation kept in an oven at 500 C for 5 hours showed less than 0.2% weight loss, while a similar piece



Figure 5.23: Dew pt. vs. time for a run with Graphite inside the furnace. Note that the average dew point level is far less than that for Alumina (Figure 5.21). Also, humps in dew point are not as prominent as in Figure 5.21.

kept at 600 C for 5 hours lost 18% weight. Thus, being exposed to 400C in air during heater burn in would do no harm to the Graphite.

Thus, a decision was made to use ¹/₄" of Graphite insulation behind the 1" of Alumina insulation on all faces of the furnace. To facilitate ease of mounting, the two pieces of insulation were sown together before mounting. On the sidewalls, in order to avoid shorting, 1.5" holes were cut out of the Graphite insulation at the locations where electrical leads passed through the walls. Figure 5.24 shows a picture of the Graphite and Alumina insulation on the front door. In case there is a leak between the insulation on the front door and the Nextel sleeved insulation on the furnace front face, hot gases from inside the furnace can reach the door and damage it. In order to prevent this from happening, ¹/₄" thick Graphite insulation which is larger than the Alumina insulation has been mounted on the door (Figure 5.24). This Graphite also serves as a
gettering agent and removes moisture from the furnace by reacting with it (See Chapter 2 for an explanation of gettering action.)



Figure 5.24: Picture showing the ¼" Graphite insulation that goes behind the 1" Alumina insulation. Also seen in front of the Alumina insulation on the door is a second layer of ¼" Graphite insulation, whose purpose is to prevent any leaking hot gas from reaching the Aluminum shell and to act as a gettering agent.

5.4. Summary

This chapter looks at the thermal design of the furnace, which involved estimating the power requirement of the large scale furnace. In addition, mechanical design of the furnace shell and some of the features on it are described. Various experiments conducted on insulation that went into the furnace are also presented. The next chapter describes the electrical, gas and safety systems of the furnace.

6 Electrical, Gas and Safety Systems

This chapter describes the electrical, gas and safety systems of the furnace. Equipment such as controllers, relays, solenoids and alarms used with the furnace were salvaged from the C.I. Hayes furnace that was talked about in Chapter 1.

6.1 Electrical systems

The large scale furnace runs in closed loop mode and has a controller that compares the desired set point with the process variable (furnace temperature), and accordingly adjusts the power supplied to the internal heaters through SCRs. The furnace controller has events that, through relays can be used to control solenoids, external heaters, internal heaters etc. In addition, the furnace has an over temperature alarm, pressure switches and a low gas flow alarm. The following sub sections describe the electrical features of the furnace. Figure 6.1 shows the convention used to represent wiring in all wiring diagrams. Most of the wires used in the furnace were salvaged form the wiring of the C I Hayes furnace that was talked about in Chapter 1. For the sake of convenience, most wire numbers (examples: 141-3, I:001 etc.) have been retained from the Hayes wiring.

 240 V AC
 110 V AC
 24V DC
 24V AC



6.1.1 Power supply

The furnace obtains its power from a 240 V AC wall power source capable of providing a maximum of 80 A. There is a manually operated safety switch on the wall that separates the furnace from the power supply. The external and the internal heaters run off this 240 V AC power supply. The furnace controls, on the other hand, operate off of 120 V AC. A transformer is used to step down the 240 V supply voltage to the 120 V required to operate the controls. The

right half of Figure 6.2 shows the wiring scheme from the transformer to the 120 V AC bus. Turning manually operated switch 157 SS operates two switches, physically located one behind the other, both of which are spring loaded 3 position switches that operate in tandem (The front switch is denoted by 157SS-F and the rear switch by 157SS-R.). To turn control power on, switch 157SS is turned to the "on" position momentarily, after which it returns to the middle position due to spring action. This causes wire 141-3 to be connected to 157-2 via 157-1. This energizes relay 157CR, which connects 141-3 to 157-1 (terminals 1 and 3). Since 157-1 coming out of terminal 3 of relay 157CR is connected together, relay 157CR remains energized even when switch 157CR returns to its middle position. To turn power off, switch 157SS is turned to the "off" position. This causes switch 157SS-R to turn to the off position, cutting off power to 157-2. This de-energizes the relay, cutting off power to the furnace controls.



Figure 6.2: Schematic showing wiring from the 240 V AC source to the 120 V AC bus used to power the controls of the furnace. Also shown in the figure is the wiring of the blower that shall be discussed in section 6.1.5.

The furnace also has a 24 V DC power supply that is powered by the 120 V AC bus. Figure 6.3 shows the wiring scheme of the 24 V DC source. The 24 V DC signal is required to trigger events from the furnace controller (a Honeywell DCP552 digital control programmer). In addition, there are two 240 V AC to 24 V AC transformers, T1 and T2, that provide power to the SCR firing cards. The wiring diagram for these is shown in Figure 6.4.



Figure 6.3: Figure showing wiring of the 24 V DC power supply (on the left). Also shown in the figure on the right is a schematic of the controller pins, to be discussed later.

6.1.2 Wiring scheme for external heaters

As explained in the previous section, the 240 V AC wall power source is capable of providing a maximum of 80 A. As stated in Chapter 5, the external heaters are each rated to supply 1 kW @240 V AC. Thus, the total current drawn when all eleven are running is 46.2 A. There are twelve internal heaters, each rated at 850 W, 120 V AC. These are connected in pairs in series across 240 V AC. Thus, the total maximum current drawn by all the internal heaters is 42.5 A. If both the internal heaters and the external heaters were to run simultaneously, then the

maximum current drawn by all the heaters would be 46.2+42.5=87.7 A, which exceeds the 80 A that the wall can supply. In order to overcome this problem, it was decided to adopt the following scheme. At the beginning of a furnace run, all eleven external heaters would be turned



Figure 6.4: Wiring diagram for the 24 V AC power supplies that are used to power the SCR firing cards, whose wiring schematic is shown on the right. The firing cards are physically mounted on to the SCRs.

on and the furnace wall would be maintained at 200 C until the dew point inside the furnace dropped to around -15 C (the internal heaters would remain off during this stage). At this point, six of the external heaters would be turned off and the internal heaters would be turned on. The internal heaters would maintain the inside of the furnace at 200 C until the dew point dropped to -25 C at which point the ramp to 1200 C would start. Under this scheme, the maximum current that can be drawn is 45.2 A (for the internal heaters) + 21 A (for the 5 external heaters), a total of 66.2 A, which is well below the maximum of 80 A.

Thus, it was decided to divide the eleven external heaters into two banks, each one controlled separately by its own relay. As seen when facing the furnace front door, Bank 1 consists of the right heater on the top face, top heater on the left face, left heater on the bottom face, bottom heater on the right face and the top heater on the back face. Bank 2 consists of the remaining heaters, including the single heater on the door. The staggering of heaters between the two banks was done in order to ensure temperature uniformity of the shell even when only one bank was turned on and the other was turned off. When carrying out a furnace run, the idea was to switch off bank 2 of the external heaters when the dew point reached -25 C, and turn the internal heaters on right after that.



Figure 6.5: Wiring diagram for relay R1 that turns bank 1 of external heaters on/off. Also see Figure 6.7.

Figure 6.5, Figure 6.6 and Figure 6.7 show a schematic of the wiring of the external heaters. Heaters of Bank 1 are hooked to the 240 V AC power supply through relay R1, while those of Bank 2 are hooked to the power supply through relay R2. The closing and opening of relays R1 and R2 is controlled by 24 V DC signals from events 3 and 4 respectively of the controller. To understand the working of the external heaters, consider external heaters 1 and 2,

labeled EH1 and EH 2 in Figure 6.7. When event 3 of the controller is active during a segment of a furnace run, the voltage across terminals 2 and 10 of relay R1 (Figure 6.5) is 24 V DC. This energizes relay R1, closing the switch between terminals 1 and 3. If the thermostats T1 and T2 (Figure 6.7) are closed (which will be the case if the shell is cooler than 200 C), contact between terminals 1 and 3 causes power to be supplied to heaters EH1 and EH2.



Figure 6.6: Wiring diagram for relay R2 that turns bank 2 of external heaters on/off. Also see Figure 6.7.

6.1.3 Controller

The controller that is used to control the furnace cycle is a Honeywell DCP 552 that is powered by the 120 V AC bus of the furnace (see list of vendors in Appendix 1 for details on the controller). The controller has two independent channels, each of which is capable of accepting its own input signal from its thermocouple and putting out a 4-20 mA output. Thus, in theory, channel 1 of the controller could be used to control the top internal heaters of the furnace while channel 2 could be used to control the bottom internal heaters. This would make it easy to turn off one zone while keeping the other one on for doing directional solidification. However, two

channels were not used to control the two heater zones, and only one channel was used instead. Details on why the two channel scheme was not used are given in Chapter 7 on furnace testing.



Figure 6.7: Wiring diagram for the eleven external heaters. Also see Figure 6.5 and Figure 6.6.

Section 6.1.4 gives details on how the top and bottom heater zones have been wired to be controlled by the output of channel 1 of the controller. Figure 6.3 shows a pin diagram of the controller. The controller's 4-20 mA channel 1 output is between pins 43 and 44.

The controller has sixteen events, each of which can be triggered during any segment of a furnace cycle. These events can be time based or process variable (furnace temperature) based. Of the 16 events, only 15 events, event #'s 2 to 16 are functional. Event 1 is damaged. Four controller events have been set up for use with the furnace. Event 2 (pin 6) turns the flow of purging gas (Argon) into the furnace off (on) and simultaneously turns the flow of Forming gas on (off). Events 3 and 4 (pins 7 and 8 respectively) are used to turn banks 1 and 2 of external heaters respectively on/off. Event 5 (pin 17) is used to turn the top internal heaters off for

directional solidification. Details of various circuits controlled by the controller events are presented in later sections.

In addition to the process controller, the furnace also has a Honeywell UDC 2000 limit controller which is used as an over temperature alarm. Details of the overtemp. circuit are given in section 6.3 on safety systems.

6.1.4 Internal heaters

As mentioned in section 6.1.2, the furnace has 12 internal heaters, each rated to deliver 850 W @ 120 V AC. Figure 6.8 shows a schematic of the heater hook up. When seen facing the front of the furnace, heater plates facing each other are connected in series. The heaters can be divided into two zones: a top zone consisting of heaters 1, 2, 3, 4, 5 and 6 and a bottom zone consisting of heaters 7, 8, 9, 10, 11 and 12. In the top zone, heaters 1 and 2 are controlled by SCR1, 3 and 4 by SCR2, 5 and 6 by SCR3. In the bottom zone, heaters 7 and 8 are controlled by SCR4, 9 and 10 by SCR5 and 11 and 12 by SCR6 (Figure 6.9).

Figure 6.10 shows a schematic of the wiring that goes into the SCR inputs. As can be seen in the figure, between the 240 V AC wall power supply and the wires going into the SCRs, there is a manual safety switch followed by a contactor, which is turned on by a 120 V AC control signal. The contactor turns on only when there is sufficient pressure in the purge and process gas tanks, the flow rate of gas through the furnace is above a preset minimum level, the furnace temperature is not above the limit set in the overtemp. controller and a manual switch is on. Details of this set up are given in section 6.3 on safety systems.

The SCRs control the power going into the internal heaters according to the control signal they receive from their firing cards (Figure 6.9). A firing card sends an output (0-100%) to its SCR according to the voltage input (1.2-6 V DC) it receives (Figure 6.4). Since the output of the controller is a 4-20 mA current, it needs to be converted into a voltage that can be fed into a firing card. Another requirement is to be able to control the power put out by heater pairs in the front, middle and rear sections (both top and bottom) of the furnace independently. Since there is greater heat loss from the furnace from near the front and rear walls as compared to the middle



Figure 6.8: Schematic showing relative positions of the internal heater plates and power connections to each plate. See Figure 6.9 for details on how the heaters are hooked to the SCRs.

section, the temperature profile inside the furnace may be non uniform. Hence the ability to control power going into heaters in different zones independently can be used to adjust the temperature profile to be as uniform as possible.

Figure 6.11 illustrates the scheme used to independently vary the control signal going into the SCRs. The figure shows a resistor, R, put across the output of the controller in order to convert the 4-20 mA signal into a 1.2-6 V signal. The calculation of R shall be explained in detail later. The resulting voltage signal is fed directly into three 10 K potentiometers, whose output goes to the firing cards of the SCRs hooked to the bottom heaters. The voltage signal from



Figure 6.9: Schematic of wiring between the SCRs and the heaters. Also seen in the figure are the control signals to the SCRs. See Figure 6.10 for schematic of wiring from 240 V AC supply to SCRs.

the output of the controller is also fed through a relay into three additional potentiometers whose output goes to the firing cards of the SCRs hooked to the top heaters. This relay, 358CR (seen in the figure) is controlled by event 5 of the controller and can be used to cut off the control signal to the top heaters, if necessary. This feature can be used to perform directional solidification by turning off the top heaters at the end of an infiltration cycle.

The value of R was estimated by considering effect of loading due to the six 10 K potentiometers, and the six firing cards, each having an input impedance of 100 K. Figure 6.12 shows a schematic of all the resistances in parallel with R across the output of the controller. Since the potentiometer resistances in series with the firing cards will be close to 10 K even after they have been adjusted to get a uniform temperature profile inside the furnace, the equivalent resistance of each pot.-firing card combination is 10K \parallel 100K =9.09K. Since there are six such pairs in parallel, their equivalent resistance is 9.09/6=1.515K. This, in \parallel with R should give 300 ohm such that a 4-20 mA current output from the controller corresponds to 1.2-6 V across R. Solving 1.515 K \parallel R=0.3 K gives R= 374 ohm. However, a 374 ohm resistor across the output did not exactly give a 1.2-6 V voltage range for a controller output of 4-20 mA. By trial and error, the right value of R was found to be 379 ohm.



Figure 6.10: Wiring from the 240 V AC supply to the SCRs. The contactor is turned on/off by a 120 V AC signal.

6.1.5 Blower

When the furnace reached its steady state operating temperature of 1200 C, it was found that some places on the outside of the bottom face of the shell reached around 260 C. This is bad from a strength point of view (the yield strength of Aluminum 5083 drops from 117 MPa at 200 C to 75 MPa at 260 C). In order to prevent the furnace shell from heating way above the design temperature of 200 C, a blower was installed on the furnace frame below the bottom face of the furnace shell (see list of vendors in Appendix 1 for details). The blower was hooked to the 120 V furnace power supply through a NC ceramic thermostat which opens at 220 C. The left half of Figure 6.2 shows the wiring diagram for the blower. Relay R3 was needed since the thermostat was NC as opposed to being NO. If blower operation is required even when the shell is cooler than 220 C, manual switch 154TS which is NC, can be flipped open to turn the blower on.

6.1.6 Gas solenoids

Event 2 of the controller is used to open/close gas solenoid valves. When event 2 is on, it energizes relay 356CR, which closes the NO valve on the Argon line while simultaneously opening the NC valve on the Forming gas line (Figure 6.13). Indicator lamp 267LT lights up when the Argon solenoid valve is open, while indicator lamp 261LT lights up when the Forming gas valve is open.



Figure 6.11: Schematic of the wiring of the potentiometers that can be used to trim the voltage signal going to the SCRs. The controller gives a 4-20 mA signal that is converted to a voltage signal using the resistor R.

6.2 Gas systems

The furnace is intended to operate under a Forming gas atmosphere for the most part, especially at higher temperatures (>900 C). However, during the soaking period at 200 C, Argon can be used to purge the furnace of water vapor. Thus, a system capable of dealing with and switching between the two gases was put in place. Figure 6.14 and Figure 6.15 show a schematic of the gas system of the furnace. The pressure regulator being used with the Forming gas tank is set to step the gas pressure down from the tank pressure to around 40 psi. A lot of this pressure drops in the moisture trap, which, due to the presence of numerous desiccant beads in it,



43

 The controller is assumed to be equivalent to a current source.
Each firing card has a 100K input impedance.

3. R in parallel with all the other resistances should be 300 ohm in order to give a 1.2 -6 V output across itself when the controller gives a current output of 4-20 mA.

Figure 6.12: Schematic of all the resistances in parallel with each other as seen from the controller's output terminals 43 and 44. The idea is to determine R that will cause the equivalent resistance across pins 43 and 44 to be 0.3 K.

Notes:

offers high resistance to gas flow. The low pressure regulator then drops the pressure down to 4 psi. The value 4 psi was chosen since the burst disc that is attached to the furnace is rated to burst at 4.5 psi, and in the event of the furnace outlet getting blocked, pressure buildup inside the furnace would not burst the disc. The low pressure regulator has good flow characteristics and consequently no droop can be observed in the outlet pressure when the flow rate through it is increased from 0 to 13 scfh (~3 volume changes an hour). The Argon gas tank on the other hand has a pressure regulator whose outlet pressure drops by around 1 psi when the flow through it is increased from 0 to 13 scfh. Although this is not too good, it is acceptable.

Both the Argon line and the Forming gas line have NO spring loaded pressure switches that close when the line pressure is >2 psig. These switches are connected in series with other switches that turn the control signal to the contactor for the internal heaters on/off. More details about this are given in section 6.3 on safety systems. The Argon line has a NO solenoid valve on

it and the Forming gas line has a NC solenoid value on it. As explained in section 6.1.6, the opening and closing of these values is controlled by event 2 of the controller.



Figure 6.13: Wiring diagram for relay 356CR which is used to close (open) the NO Argon solenoid while simultaneously opening (closing) the NC Forming gas solenoid.

The idea is to be able to automate events such as switching from Argon to Forming gas after the purge phase of the furnace run and switching from Forming gas to Argon before the melting point of the infiltrant (to avoid dissolution of Hydrogen in the infiltrant melt) to occur through a furnace program. The wiring of the solenoid valves in Figure 6.13 ensures that both Argon and Forming gas can never be turned on simultaneously. If this were to happen, excess flow would occur through the flow meter (range of operation: 0-30 scfh) and damage it.

The two needle valves seen in Figure 6.14 can be preset at different openings so as to pass different flow rates of gas through each (flow rate of Argon, which is used for purging, will typically be lesser than that of Forming gas used during the rest of the process). The idea of having check valves after the pressure regulators is to prevent one gas from filling up the tank of the other in case the solenoid valves fail.

Figure 6.15 shows the gas line coming out of the furnace shell. Positive pressure (~ 4-5" of water) is maintained inside the furnace using the modified flow meter shown in the figure. The device is a flow meter with no needle valve. The weight of the float inside the flow meter was changed (by changing the material of the float), until the weight was such that a furnace



Figure 6.14: Schematic of the gas system used in the furnace. There are two possible atmospheres the furnace has been set up to run in: Argon and Forming gas.

pressure of around 4" of water was developed with 13 scfh flowing through the furnace. The advantage of this set up over using a needle valve is that there is a lower chance of the outlet getting clogged, as the float floats high above the gas inlet of the modified flow meter. Also, since the flow meter is transparent, a quick indication of normal flow through the furnace can be obtained by just looking to see if the float is floating in the tube.

There is a dew point sensor on the line after the modified flow meter. This is followed by tubing to an exhaust vent. This tubing is long enough (>1m) to prevent diffusion of air from the atmosphere from affecting dew point readings.

6.3 Safety systems

The idea behind the safety systems is to protect the furnace and the parts being processed in it in the event of things going wrong. As a first step, the furnace shell has an overpressure burst disc, is grounded and has fuses at appropriate places. The furnace has been provided with a



Figure 6.15: Schematic of the gas flow line from the furnace outlet to the exhaust duct. Honeywell overtemp. controller (details in list of vendors in Appendix 1) which is set to cut off power to the internal heaters (and thus prevent a furnace meltdown) if the temperature inside the furnace exceeds 1250 C. This overtemp. controller gets its temperature input from an over temperature thermocouple mounted into the furnace through a fitting on the rear wall. The furnace also has a thermocouple that measures the temperature of the bottom heater zone, and displays it on Channel 2 of the furnace controller. Thus, the furnace has three thermocouples in all, including the control thermocouple that is connected to channel 1 of the controller. The Argon and Forming gas lines have been provided with pressure switches that cut off power to the internal heaters if the pressure in either line goes below 2 psi (Figure 6.14). In addition to this, the furnace has a low flow alarm that cuts off power to the internal heaters if the flow of gas through the furnace is below a set limit. These features ensure that the furnace will never run without gas, and thus protect the quality of the parts being processed in the furnace.



Figure 6.16: Schematic of the wiring of safety switches in series. The switches carry the 120 V AC control signal to the contactor, 258CON, that turns on power to the internal heaters.

Figure 6.16 shows a schematic of the safety switches wired in series. They carry the 120 V AC signal from 157-1 to I:001 that turns the contactor (258CON) for the internal heaters on. The flow alarm circuit board's terminals are shown in Figure 6.17. The flow alarm sensor body is mounted on to the flow meter, and the circuit board is mounted on the furnace frame. Switch S2 of the overtemp. controller is NO and is hooked to a buzzer. The wiring for the buzzer is shown in Figure 6.18. When the overtemp. condition is met, I:003 gets live and energizes relay 279CR. This connects wire 157-1 to 190-1, which being connected to 190-2, activates the buzzer. 194PB is a spring loaded pushbutton switch (NO) that is used to silence the alarm buzzer. Pushing it causes a momentary contact between 194-1 and 190-1. This energizes relay 194CR, which cuts off contacts 1 and 4, silencing the alarm. Lamp 194PBLT remains on as long as the overtemp. alarm condition is on.



Figure 6.17: Schematic of the flow alarm's pins. Switch S3 is normally closed. The flow alarm sensor body is mounted on the flow meter, while the circuit board is mounted on the furnace frame.



194 PB: NO spring loaded pushbutton switch. If the alarm condition is on, pushing 194PB causes the alarm to silence and lamp 194PBLT to glow.

190AH: Alarm buzzer

Figure 6.18: Wiring diagram for the overtemp. buzzer alarm which gets activated by an alarm signal from the overtemp. controller. The spring loaded pushbutton, when pressed, silences the alarm.

6.4 Conclusion

This chapter presented the electrical, gas and safety systems of the furnace. Detailed wiring diagrams were provided for all the circuits in the furnace. Details of the furnace gas system were also presented. The next chapter describes furnace testing and the procedure that was used to obtain optimal quality parts using the furnace. It also describes the results of temperature profiling experiments that were carried out inside the furnace.

7 Furnace Testing

This chapter describes the steps leading to the successful firing of the furnace to obtain oxide free parts. It also describes temperature profiling experiments aimed at obtaining a uniform temperature profile along the furnace depth by independently adjusting the power going into the front, middle and rear heaters using trim potentiometers, if necessary.

7.1 Initial furnace runs

The first time the furnace was fired, the controller was operated in manual mode and the ramp to 1200 C was carried out with 100% power going to all the heaters. This procedure, of ramping to 1200 C at full power is recommended by the heater manufacturer, and is required to "transform" the APM wire of the heaters. As the furnace temperature approached 1200 C, the power going to the heaters was turned down by trial and error to a value (~78%) that could maintain 1200 C in steady state. The furnace was then held at 1200 C for four hours to complete the "transformation" of the APM wire. Air flow through the furnace was maintained at 2 volume changes an hour (9 scfh) throughout this process. This was done for two reasons: first, the "transformation" of the APM wire needs to happen in air, and second, the presence of air would ensure complete burning off of the polymer that was sprayed on to the Alumina insulation to make its mounting into the furnace easy (see Chapter 5). When the furnace was opened after this run, the Inconel pins, mounts and mesh inside were found to be oxidized since the furnace had been run in air. However, this oxide got reduced in subsequent runs that were carried out in a Forming gas (5% Hydrogen, 95% Argon) atmosphere.

The next furnace run involved auto tuning the furnace controller at three temperatures in order for it to determine the right PID variables for optimal control at these temperatures (procedure is described on page 9-6 of the controller programming manual). The three temperatures chosen for auto tuning were 500 C, 1000 C and 1200 C, and the corresponding PID parameters were automatically written by the controller into PID constants PID1, PID2 and PID 3 respectively. Thus, while programming any furnace cycle, PID1 should be chosen for segments with set points (SP's)<=500 C, PID2 for those segments with 500</p>

Two tests were conducted to test if the controller was indeed doing a good job controlling the furnace. These involved looking at the response of the furnace to a step change in set point. In the first experiment, the furnace was allowed to reach steady state at 1180 C, after which the furnace set point was programmed to ramp up to 1200 C in 1 second. Figure 7.1 shows the response of the furnace to this step change. In the next test, the furnace temperature was allowed to reach steady state at 800 C before bumping the set point down to 780 C in 1 second. Figure 7.2 shows the furnace response to this step change. The absence of any major overshoot in furnace temperature in either case illustrates good control by the controller.





7.2 Determining an optimal furnace run

This section describes experiments that were conducted in order to determine a furnace ramp scheme that would yield oxide free parts in the shortest possible furnace cycle.

7.2.1 Furnace runs for obtaining oxide free parts

Having auto tuned the controller, the next step was to sinter SS420 powder in the furnace and obtain an oxide free sintered part. As can be recalled from Chapter 2, the procedure followed while running the small scale furnace was to let the furnace soak at 200 C (with the internal and external heaters both set at 200 C) until the dew point inside got below -25 C, and then ramp the furnace to 1200 C.



Figure 7.2: Graph showing the response of the furnace to a step change in set point from 800 C to 780 C. The absence of any significant overshoot in the response shows good controller action.

A similar procedure was followed with the large scale furnace. As was explained in Chapter 6, due to a limit on the amount of current that can be drawn from the wall (maximum of 80A), all the external heaters cannot be run simultaneously along with the internal heaters, unlike in the small scale furnace where this could be done. Hence, the following procedure was used. All eleven band heaters were turned on (with the internal heaters kept off), and the furnace was purged with Forming gas at 3 volume changes an hour (13-14 scfh) until the dew point inside the furnace dropped to around -10 C to -15 C. Then, six of the external heaters were turned off (using a controller event) and the internal heaters were turned on. The controller set point was set



Figure 7.3: Dew point curve for a furnace run (run 1) conducted using the scheme described in the text. Note that the purging time (regions 1 and 2) is around 4 hours. Compare this with run 2 in Figure 7.5, where the purge time is 3 hours.

at 200 C, and the furnace was allowed to soak at 200 C until the dew point got down to -25 C. At this point, the furnace was ramped up to 1200 C at 5 C/min. Figure 7.3 shows the dew point curve that was obtained the first time the furnace was run according to the above mentioned scheme (call this run 1). It can be seen that the nature of the dew point curve in region 3 is similar to that seen in runs with the small scale furnace (see Chapter 2). The external strip heaters were turned off at the end of this run. Run 1 yielded an oxide free sintered SS part (Figure 7.4). It is worth noting that clean parts were obtained using just 3 volume changes an hour of Forming gas as opposed to the 5 volume changes an hour used in the small scale furnace. After the furnace cooled down to room temperature at the end of the run, it was opened for 40 minutes, loaded with another SS420 sample and the previous run was repeated (call this run 2). Figure 7.5 shows the dew point curve for run 2. Two things can be noted by comparing Figure 7.3 with Figure 7.5. First, in run 2, it took just 185 minutes before the dew point was low enough to start the ramp to 1200 C as opposed to the 242 minutes taken in run 1. This is a saving of almost 1

hour in purging time. The second difference between the two runs is that the dew point levels in run 2 are lower



Figure 7.4: Picture of the SS420 part that was successfully sintered in the furnace using the scheme described in section 7.2.1. The dew point curve for this run is shown in Figure 7.3.

than those in run 1. The difference in the dew point curves could be due to the fact that the insulation did not have enough time to soak up water vapor from the atmosphere, as the furnace was opened for just 40 minutes between the runs.

To get an idea of how many volume changes of ambient air would be needed to saturate the furnace insulation, the following calculation was performed. The total mass of Alumina insulation inside the furnace was calculated to be ~3 kg, while that of the Graphite insulation was estimated to be ~ 0.75 kg. Alumina absorbs 5% moisture by weight and Graphite absorbs 2% moisture by weight. Hence, the total moisture that would be needed to saturate the insulation inside the furnace with moisture is 165 gram. Under normal room conditions (25 C, 50% RH), a psychrometric chart shows that 1 gram of water vapor is contained in 0.77 cubic meter of air. Thus, the volume of air that would contain enough water vapor to saturate the insulation is 13 cubic meter. The volume of the furnace is 4.5 cubic feet, which is roughly 0.127 cubic meter. Thus more than 100 volume changes of air would be needed to saturate the insulation. This points to the fact that if the furnace is opened for a limited amount of time between runs, not much moisture will be absorbed by the insulation. In addition, if the strip heaters of the furnace are kept on all the time, then the insulation will be at around 200 C all the time and this will further reduce the amount of water vapor absorbed from the atmosphere each time the furnace is opened. To test this assumption, two back to back runs (call them run 3 and run 4 respectively) were conducted in the furnace. The furnace was left sitting open for five hours after a previous run that had been performed just before run 3. The purge time (time from beginning of run to the point when the dew point first reaches -25 C) during run 3 was found to be 243 minutes. Unlike in other runs, the strip heaters were not turned off at the end of run 3, but were left on while the



Figure 7.5: Dew point curve for run 2, conducted right after run 1 whose dew point curve is shown in Figure 7.3. The furnace was opened for 40 minutes after it cooled down to room temperature at the end of run 1, before starting run 2. Note that the purging time for this run is around 3 hours as opposed to 4 hours for run 1.

furnace cooled. When the furnace cooled down to 195 C (steady state internal temperature reached with all eleven external heaters on), it was opened (the strip heaters were left on) for exactly 10 minutes and then closed again for performing run 4. For this run, the purge time was found to be just 124 minutes. This represents almost a 50% reduction in purge time over that in run 3.

7.2.2 Other ramp schemes

Looking at Figure 7.3 and Figure 7.5, it can be seen that part of the reason it takes a long time \sim 4 hours for the dew point inside the furnace to reach -25 C is that the inside of the furnace heats up very slowly (at about 1 C/min) when the internal heaters are off and only the external heaters are heating the furnace. The furnace heats up slowly because heat from the external

heaters has to travel through the furnace insulation and heat up the significant thermal mass of the furnace, including that of the internal heater plates. Ideally, one would want to turn both the



Figure 7.6: Dew point curve for run 5. The bottom zone of internal heaters was kept on with a set point of 200 C during the purge in order to quickly heat up the inside and speed up the moisture removal process.

top and bottom internal heater zones (with a set point of 200 C) and the external heaters on simultaneously, to get the fastest expulsion of moisture. However, this cannot be done due to a restriction on the amount of current that can be drawn from the wall. Hence, a run was carried out (call it run 5), during the purging phase of which the bottom internal heaters were turned on with a set point of 200 C, all eleven external heaters were turned on and the top internal heaters were kept off (see appendix 3 for details of a furnace program similar to one used in this run). The maximum current drawn under this scheme was 67.5 A, well below the 80 A limit imposed by the wall power supply. The top heaters were conveniently turned off using event 5 of the controller. Figure 7.6 shows the dew point curve obtained during run 5. It can be seen that the purge time is around 184 minutes, a significant reduction from the 242 minutes in run 1, performed without turning on any internal heaters during the purge (Figure 7.3). One thing that can be noted from Figure 7.6 is that the Forming gas coming out of the furnace is saturated

during the period marked 1'. This shows that the limitation on the rate of moisture removal from the furnace is posed by the gas flow rate. Clearly, 13 scfh is not enough to remove moisture from the furnace fast enough to avoid saturation. A possible solution to this problem is to use a higher



Figure 7.7: Dew point curve for run 6. The gas flow rate in this run was 30 scfh as opposed to the 13 scfh used in run 5. Note that the purge time in this run is 122 minutes, more than 60 minutes lower than that in run 5, whose dew point curve is shown in Figure 7.6.

gas flow rate. As noted in Chapter 6, the maximum gas flow rate that the furnace's flow meter can handle is 30 scfh. Thus, in the next furnace run (call it run 6), the flow rate of Forming gas was set at 30 scfh and the same furnace cycle as that used in run 5 was followed. Figure 7.7 shows the dew point behavior for this run. It can be seen that the purge time is just a little over 120 minutes, more than a 60 minute reduction over the 184 minutes seen in run 5, that used 13 scfh of Forming gas. Also, the overall dew point levels in run 6 are lower than those in run 5. Figure 7.8 is the picture of a part that was successfully infiltrated in this furnace run.

Thus, it can be seen that turning on the bottom internal heaters during the purge phase and increasing the flow rate to 30 scfh helps in significantly lowering the purge time compared to the scheme followed in section 7.2.1. Even greater reduction in purge time may be obtained if, in addition, the furnace shell is always maintained at 200 C by keeping the external heaters on between runs and the furnace door is mostly kept closed except while loading/unloading parts in and out of the furnace.



Figure 7.8: Picture of a SS 420- Copper part that was infiltrated successfully during run 6. The dew point curve of the cycle used in run 6 is shown in Figure 7.7. The part was suspended inside the furnace using the 1/8" diameter Moly. rod seen in the picture, and dipped into the copper melt only after all the copper had melted and steady state had been attained at 1200 C.

7.2.3 Dew point sensing

As can be gathered from Chapter 2 and sections 7.2.1 and 7.2.2 of this chapter, sensing the dew point inside the furnace is a critical part of every run. The method currently being used is to "hold" the furnace run when it is in the purging segment, and wait for the dew point to get to -25 C before manually resuming the ramp to 1200 C. This method is cumbersome since it

requires the furnace operator to monitor the dew point at regular intervals of time. Also, the dew point sensor currently being used is very basic and only has a digital readout (see list of vendors in Appendix 1 for details on the dew point sensor).

There are two ways of getting around the problem of having to monitor dew point manually. The first is to simply assume that a 3 hour purge at 200 C will be sufficient to attain a dew point of -25 C inside the furnace, and program the furnace accordingly. The other solution is to automate the entire run/hold process. A few general comments which might be helpful while designing such a scheme follow. A feature of the furnace controller that could be used is the fact that the "hold" and "run" functions of the controller can be operated using momentarily applied external switch inputs (see pages 9-8 and 9-9 of the controller manual for details). The "hold" switch could be closed momentarily at the arrival of the "hold at 200 C" segment of the furnace run, indicated by, say, a short furnace time based event. Using a dew point sensor that gives a 4-20 mA output corresponding to a given dew point range, and a current sensing relay with a set threshold current, a dew point of -25 C can be detected. The current sensing relay could then be used to close the "run" switch which just needs momentary contact. Once this has been done, even when the dew point rises above -25 C as the furnace ramps up and the current sensing relay switch opens, it will not have any effect on the "run" condition, which can only be overridden by a hold signal.

7.3 Two zone control scheme and temperature profiling

7.3.1 Experiments with two zone control

As explained in Chapter 6, the furnace controller has two independent channels that can accept inputs from separate thermocouples and give out separate 4-20 mA outputs. During initial stages of furnace testing, SCRs controlling the top heaters of the furnace were wired to receive their input from channel 1 of the controller while SCRs controlling the bottom heaters received their input from channel 2. The idea was that this would make it easy to shut off the top heaters in order to perform directional solidification, as it would simply involve programming channel 1



Figure 7.9: Picture of the furnace fitted with the door used for temperature profiling. and channel 2 of the controller separately. This method was tried out, by hooking channel 1's input to a thermocouple located in the top heater zone and channel 2's input to a thermocouple located in the bottom heater zone. However, when the furnace was fired, it was found that when steady state was attained at 1200 C, the top thermocouple read 1200 C, while the bottom one read 1193 C. This was not surprising since the tolerance of type K thermocouples at 1200 C is +/-9 C. However, what was of concern was the fact that the bottom heaters were running at 100% power while the top heaters while running at only 53% power. Clearly, channel 2 of the controller, which controlled the bottom zone, was trying its best to get the temperature read by its thermocouple to 1200 C by making the bottom heaters put out 100% power (in fact the bottom zone of the furnace was measured to be at 1211 C (+/-9 C) using independent profiling thermocouples through the front door. Moreover, when the bottom thermocouple was raised to the same height as the top one, it was found to read 1194 C while the top one read 1200 C. All this points towards the advantage of using a low error, custom thermocouple calibrated for use at 1200 C to control the furnace.). Since the furnace heating zones are highly coupled (each location in the hot zone has a good radiation view factor with respect to the other), the bottom heaters putting out 100% power caused the top heaters to cut back their power to 53%. (If this happens over an extended period of time, it will result in a difference between the lives of the top and bottom heaters.)



Figure 7.10: Schematic of a section through the furnace, as seen from the side. Profiling was done for x=0 to x=18.5" along the furnace depth at each of the six ports shown in Figure 7.9. The front, middle and rear heater plates are respectively 6", 8" and 6" long and roughly extend from the front insulation to the rear insulation.

To verify the fact that the top and bottom heater zones were indeed coupled, the furnace control was changed to manual at 1200 C and the following pairs of power output going to the top and bottom heater zones were tried out: (53%,100%), (63%,90%), (68%,85%) and (73%,80%). The idea was to keep the total power going into both zones combined roughly constant, and only vary the its distribution between the top and bottom heaters. In a highly coupled system, this should not have an effect on the temperature recorded by either the top or the bottom thermocouple. At each power setting, temperatures read by the two thermocouples were recorded after steady state was attained. The following values were recorded by the top and bottom thermocouples respectively, corresponding to the power output % pairs given above: (1200,1193), (1195,1184), (1195,1181) and (1198,1181). Clearly, the difference between the readings of the two thermocouples is not significant, if the +/-9 C tolerance of each thermocouple is taken into account (A point worth mentioning is that the two channels of the controller were

found to read different temperature values when connected to the same thermocouple, probably due to a difference in their A/D converters. This further adds to the argument against using two zone control.).



Figure 7.11: Schematic showing the relative location of the thermocouple ports on the front door.

This proved beyond doubt that two zone control was not worth it, since the top and bottom zones of the furnace are highly coupled. Hence, a single controller channel, channel 1 was used to control both heater zones using the thermocouple in the top heater zone. (The wiring for this scheme is explained in detail in Chapter 6.)

7.3.2 Temperature profiling

As mentioned in Chapter 5, the furnace has a door that can be used for temperature profiling. Figure 7.9 shows a picture of the furnace fitted with the profiling door. Six thermocouple feedthroughs can be seen on the door. These are 3/16" bored through pipe fittings whose bore was drilled to a diameter of 0.205". This was done to prevent jamming of the 0.1875" outer diameter Mullite thermocouple sheath inside the fitting due to thermal expansion.

The Mullite sheath is 24" long and has two holes in it, each of which has a diameter of 0.060". Each wire of the type K thermocouple that passes through the sheath has a diameter of 0.032". To prevent inaccuracies resulting from differences in different thermocouples of the same type,



Figure 7.12: Temperature profiles obtained along the depth of the furnace at each of the six ports shown in Figure 7.11. Note that the temperature is in a 5 C band over the length of the hearth plate.

the same thermocouple was used to profile the furnace depth at all six locations in the door. Figure 7.10 shows a schematic of a section of the furnace looking from the side. Figure 7.11 is a schematic showing the relative locations of the thermocouple ports (numbered 1 to 6) on the profiling door as seen from the front. Temperature profiling was carried out at each of the six ports. At each port, temperature was recorded at 11 locations along the furnace depth, from x=0 to x=18.5". When the thermocouple is at x=18.5", due to the clearance between the sheath, (which has a diameter 0.1875") and the 0.205" diameter hole in the fitting in the door, the tip of the thermocouple describes a circle of radius 0.5". This introduces a slight uncertainty in the exact location at which temperature is measured. Figure 7.12 is a graph showing the temperature profile along the depth of the furnace at each of the six ports, 1 to 6. It is important to note that the standard type K thermocouples used for profiling have an accuracy of +/-0.75%, which

translates to an accuracy of $\pm 0^{\circ}$ when measuring a temperature of 1200 C. The hearth plate extends form x=2.5" to x=16.5". It can be seen that the temperature in this zone is in a 5 C band. This represent very good temperature uniformity in the furnace hot zone.



Figure 7.13: Temperature profiles obtained at port 2 as part of the trimming experiment carried out to try and smooth out the temperature profile inside the furnace.

Although this was not very crucial, (in view of the excellent temperature uniformity already existing in the furnace) a few experiments were conducted to try and adjust the power going into the front, middle and rear heater zones in such a way as to smoothen the profiles shown in Figure 7.12. As discussed in Chapter 6, potentiometers were wired to the controller output to independently vary the control signal (and thus the power) going to the top and bottom heaters of the front, middle and rear zones, and thus control the relative temperature profile inside the furnace.

In steady state, with all potentiometer knobs turned to 100%, the controller put out 77% power to all the heaters. The temperature profile along the furnace depth at port 2 for this setting is shown in Figure 7.13 (call this the base case, labeled 1 in the figure). Note that the temperature between x=2.5" and x=16.5", the hearth plate zone, is in a 3 C band. Also note that the temperature drops off rapidly towards the front of the furnace (x=0). It also drops a little towards

the rear of the furnace (x=18.5"). In order to "lift" the profile at the front and the rear, the potentiometers controlling the middle heaters were adjusted to the 90% setting. This would raise



Figure 7.14: Temperature profile along the height of the furnace, measured using a port in the middle of the top face of the furnace. Note the tight 2 C band the temperature is in over a 14" height.

the power going to the end heaters relative to the middle ones. In steady state under this setting, the controller output went up to 80.5% and the resulting temperature profile is labeled 2 in Figure 7.13. In this case, the power going to the front and rear heaters is 80.5% and that going to the middle heaters is 0.9*80.5% = 72.45%. Note the rise in temperature in the front and rear zones over the base case. Also, the overall variation in temperature over the hearth plate zone (x=2.5" to x=16.5") reduced to 2 C from 3 C in the base case. However, the steep fall in temperature as x tends to 0 still exists. To try and rectify this, the potentiometer knobs of the middle heaters were turned down further from 90% to 80%. In steady state, the controller output went up to 83%. In this case, the power going to the front and rear heaters is 83% and that going to the middle heaters is 0.8*83% = 66.4%. The resulting temperature profile is labeled 3 in Figure 7.13. Note that the overall temperature in the hearth plate zone goes up a couple of degrees. Interestingly enough, the temperature as x tends to 0 actually decreases as compared to
the previous cases. Somehow, the extra power going into the front heaters ends up heating up the middle zone of the furnace instead of the front zone.

These experiments showed that trimming did have a small impact on the relative temperature profiles in the sense that it tightened the band over which the temperature varied over the length of the hearth plate from 3 C to 2 C. Having said this, however, it should also be noted that the temperature profile obtained inside the furnace with all the trim pots set at 100% is also very acceptable for processing parts without distortion.

The next experiment involved obtaining a vertical temperature profile inside the furnace. Temperature was recorded starting at y'=1" (see Figure 7.10 for details of coordinate system) from the hearth plate to y'=14.75", which is 1" below the top insulation. This profile was obtained using the QF40 fitting in the middle of the top face of the furnace. Figure 7.14 shows the temperature profile obtained. Note that the temperature is in a tight 2 C band over a height of around 15" starting at the hearth plate. This is excellent temperature uniformity.

In order to get a sense of the vertical temperature gradient that could be induced inside the furnace for performing directional solidification, the furnace was switched to manual mode when steady state was attained at 1200 C. Then, the top heaters were shut off and the bottom ones were given 100% power. The furnace started to cool, and in steady state (bottom thermocouple reading 1013 C), a temperature difference of 23 C was recorded between thermocouples in the top and bottom zones, which were separated vertically by about 7" (a temperature gradient of a little more than 3 C/inch). This further demonstrates the fact that the top and bottom zones of the furnace are highly coupled (a point made in section 7.3.1.) and that large temperature gradients cannot be obtained inside the furnace. However, is should be noted that the presence of a large part inside the furnace shall decrease cross radiation from the bottom heaters on one face to the top corner of the opposite face, possibly giving rise to a higher temperature difference between the top and the bottom zones of the furnace.

7.4 Summary

If the furnace power supply has a high enough current capacity (~90 A) to permit running all the internal and external heaters simultaneously, then it is recommended that the furnace be purged with the internal heaters set at 200 C, and all the external heaters turned on. Otherwise, the schemes identified in this chapter can be used to purge the large scale furnace in order to obtain oxide free parts.

It was shown in this chapter that excellent temperature uniformity exists inside the furnace, both from top to bottom and along the depth in the region where the hearth plate is located. Reasons for not using two zone control in the furnace were discussed. The results of some experiments examining the effect of trimming the power going into heaters in different zones on the furnace temperature profile were also discussed. The next chapter, which is the final chapter of this thesis gives a summary of the achievements of this work, and recommendations for future work using the large scale furnace.

8 Conclusion and Future Work

This chapter summarizes the achievements of this work and talks about future work that can be carried out using the large scale furnace that was successfully built and tested as part of this work.

8.1 Summary of achievements

As mentioned in Chapter 1, the primary motivation behind this work was to develop a post processing furnace for 3D printed parts that could be scaled up economically. As a first step, the small scale furnace previously developed at MIT was optimized to yield good quality, oxide free sintered and infiltrated parts. The important thing learnt from the small scale furnace was that it in order to get oxide free parts, it was necessary to heat the furnace shell from the outside and allow the furnace to soak at 200 C until the dew point inside got to a low value like -25 C, before ramping the furnace to 1200 C.

Having learnt about the process and obtained design parameters from the small scale furnace, a large scale furnace was constructed. This furnace has a rectangular cross section and is more than 20 times bigger in volume than the small scale furnace, which had a circular cross section. This furnace has an Aluminum shell, uses Alumina and Graphite insulation and uses Kanthal heating elements embedded in high Alumina content heater plates to heat the inside of the furnace (12 internal heaters for a total of 10.2 kW). It is designed to operate in a Forming gas or Argon atmosphere. In addition to the internal heaters, it has eleven strip heaters on the shell (a total power of 11 kW) to expel moisture from the insulation during the purging phase of a furnace run. The furnace has a Honeywell controller that uses PID control to ensure the desired operation of the furnace. The controller has events that are used to control the switching of the strip heaters and gas solenoid valves.

SS420 parts were successfully sintered and infiltrated in this furnace using a furnace cycle similar to that used in the small scale furnace. In fact, clean parts were obtained in the large scale furnace using just 3 volume changes an hour of Forming gas as opposed to the 5 volume changes an hour flow rate used in the small scale furnace. Temperature profiling done inside the furnace showed excellent temperature uniformity inside. It was found that the temperature along the depth of the furnace was in a 5 C band over the 14" length of the hearth plate. Also, the

temperature was found to be in a 2 C band over a height of 15 inches going upward from the hearth plate.

The successful scaling up and operation of this furnace proves the concept of economical scalability. However, one of the major problems with the large scale furnace design as it stands is that of assembly of internal heaters into the furnace. As may be recalled from Chapter 5, the furnace had originally been designed to have 1" of insulation on all faces. However, an additional 0.25" of Graphite was used on all faces to decrease heat loss from the furnace, and an Inconel mesh was used on the front, top and rear faces to prevent the insulation from falling apart. The presence of extra insulation and Inconel mesh on the top face of the furnace made mounting the top heaters onto the furnace side walls very difficult since there was very little room at the top for maneuvering the heaters into their Inconel supports. Reaching into the interior of the furnace through the front door in order to mount the four heaters located towards the furnace was found to be hard. Similarly, mounting insulation onto the top face of the furnace of the furnace of the furnace and to be difficult since it was not easy to reach the Inconel supporting pins located towards the rear of the furnace.

If the existing design of the large scale furnace is scaled up, it will be even harder to mount insulation and heaters into the furnace, primarily because of difficulty accessing the innards of the furnace through the front door. To overcome this problem, it is suggested that a SS cage housing the insulation and heaters be assembled outside the furnace and then be mounted into the furnace as a single unit.

A second issue that was studied as part of this work was the effect of directional solidification on porosity in infiltrated parts. The small scale furnace was used to conduct directional solidification experiments on SS420 parts infiltrated with Bronze. These involved inducing a temperature gradient in a part (using a jet of Argon striking the part on the top) at the end of an infiltration cycle. By comparing two extreme cases, directional solidification, in which the top of a solidifying part is kept cooler than the bottom, with reverse directional solidification, in which the bottom of the part is kept cooler than the top as it solidifies, it was established that directional solidification reduces porosity in parts. A switch that is controlled by a furnace controller event has been provided in the large scale furnace to shut off the top heaters in order to conduct directional solidification in cooling parts.

The economics of requiring two separate furnaces for post processing, one for debinding and the other for sintering was addressed in Chapter 1. An apparatus for carrying out debinding and sintering in the same furnace run without contaminating the furnace was designed and successfully tested in the small scale furnace. This apparatus can be scaled up for use in the large scale furnace, and will eliminate the need for using two separate furnaces for post processing, resulting in a halving of capital cost.

8.2 Recommendations for future work

As mentioned in the previous section, the large scale furnace has a switch that can be used to cut off power to the top heaters. This feature can be used to perform directional solidification at the end of an infiltration, just before cool down. The effect of directional solidification on porosity that was identified during experiments conducted in the small scale furnace can be tested using the large scale furnace.

The furnace has a number of QF40 flange fittings on the top face. One of these can be used as a passage for the actuator of a gating mechanism that can be used to perform gated infiltrations. These experiments can be carried out to learn more about the effect of gated infiltration on porosity in infiltrated 3D printed parts. In addition, the debinding apparatus that was successfully tested in the small scale furnace can be scaled up and be used to debind and sinter green parts in a single run in the large scale furnace.

Appendix 1: List of Vendors

S.no.	Item name	Item specifications	Vendor
1	Alumina insulation	Alumina blanket type AB, 95% Alumina content	Zircar Products Inc. 110 North Main Street PO BOX 458 Florida, New York 10921-0458 Phone: 914 651 4481
2	Brass Connector	Connectors to connect wires from SCR to the Brass tubes coming out of the furnace	MIT Central Machine Shop See part drawing in Appendix 2
3	Braze alloy	Eutectic alloy: Ag 72% Cu 28% 0.062" diameter SB72 wire	Prince/ Izant Company 12333 Plaza Drive Cleveland, OH 44130
4	Braze Flux	Ultra black flux	Phone: 216 362 7000
5	Burst disc	Part # 1MA1X2X5X72 2" Graphite disc, burst pr. 5psig@ 22 C and ~4.5 psig @ 200 C	Zook Enterprises, LLC 16809 Park Rd., Chagrin Falls OH, 44022 Phone: 440 543 1010
6	Ceramic tubes	Mullite tubes for hearth plate supports, heater location and electrical insulators, see part drawings in Appendix 2	East Coast Sales Co. Inc. 554 North State Road Briarcliff Manor, NY 10510 Phone: 914 923 5000
7	Contactor	Turns power to internal heaters on/off Is controlled by a 120 V AC signal from the controller Part# AB 100 A60 ON D3 Max current : 60 A @ 240 V AC	Allen Bradley 1201 South Second Street Milwaukee, WI 53204 Phone: 414 382 2000 Local retailer, Boston area: Eagle Electric Phone: 781 302 2000
8	Controller	DCP 552 Digital Control Programmer Two Channel Model HW DCP 552 E20 200	Honeywell Inc. Industrial Controls Div. 1100 Virginia Dr., Ft. Washington, PA 19034-3260 Phone: 215 641 3000 Customer Service: 800 423 9883
9	Dew point sensor	Part# E-37450-04F Thermohygrometer with dew point	Cole Parmer Instrument Co. E. Bunker Court, Vernon Hills IL 60061 Phone: 800 323 4340

10	Flow meter	Waukee flow meter, type MPX-2, 0-30 CFH Argon, 5psig	Waukee Engineering Co. Inc. 5600 West Florida Avenue Milwaukee, WI 53218
11	Flow alarm	Waukee flow alarm, model ML	Phone: 414 462 8200
12	Furnace shell	Welded Aluminum box with various fittings. See part drawings in Appendix 2	NU VACUUM SYSTEMS INC. 23 Joseph Street Kingston, MA 02364 Phone: 781 585 4371
13	Fuses	External heaters: BUSS MDL 12, 12 A, 250 V time delay fuses Before/after transformer : BUSS FRNR-7, 7A, 250V dual element, time delay fuses	Newark Electronics 59 Composite Way, Lowell MA, 01851-5150 Phone: 978 551 4300
14	Hearth plate	Graphite ISO 63 grade, see drawing in Appendix 2	Graphite Engg. & Sales Co. 712 Industrial Park Drive PO Box 637 Greenville MI 48838-9984 Phone: 800-472-3483
15	G-10 heat shields	Garolite G-10 sheet 1/8" thick, 2' by 2'	Mc Master Carr PO Box 440 New Brunswick, NJ 08903-0440 Phone: 732 329 3200
16	Graphite insulation	Graphite felt, GH grade, 1/4" thick	Fiber Materials Inc. 5 Morin Street Biddeford, ME 04005-4497 Phone: 207 282 5911
17	High temp. wiring	For wiring external heaters. Model #: HTMG-1CU318S/C 18 AWG Mica glass insulated ultra high temp. wire rated for use at 450 C	Omega Engineering Inc. One Omega Drive PO Box 4047 Stamford, CT 06907-0047 Phone: 800-TC-OMEGA
18	Inconel pins,mounts, washers	For mounting insulation and heaters inside the furnace. See part drawings in Appendix 2	Axis Technologies 27 Industrial Avenue Chelmsford, MA 01824 Phone: 978 250 9909
19	Inconel mesh	Part # : Inco 601, 0.063" wire diameter 2 mesh	Alliant Metals Inc. 134B, Route 111, Hampstead, NH 03841 Phone: 800 543 1453

20	Internal heaters	FPH207-S-L-APM heaters as per drawings in Appendix 2	Thermcraft Inc. PO Box 12037 Winston-Salem, NC 27117 Phone: 336 784 4800
21	Moisture trap	Model # : 8012-4	BOC Gases PO BOX 3053 Edison, NJ 08818-9869 Phone: 800 892 7706
22	Pressure regulator	Part# ADS-SSC-5, low pressure regulator, 1.1-5 psig outlet pressure	Airgas Northeast 17 Northwestern Drive PO Box 1647 Salem, NH 03079 Phone: 603 890 4600
23	Nextel	Nextel 440 BF20 fabric Nextel 440 BT30 thread heat cleaned (sizing removed)	Thermal Ceramics 2730 INDL PKWY Elkhart, IN 46516-5401 Phone: 330 562 8890
24	Overtemp. Controller	Universal digital controller, limit control model Model# HW CD200H-2-000-1000000T*	Honeywell Inc. Industrial Controls Div. 1100 Virginia Dr., Ft. Washington, PA 19034-3260 Phone: 215 641 3000 Customer Service: 800 423 9883
25	Polymer spray	Poly Acrylic Acid, 35 wt% solution in water. Average Mol. Wt. 250000 Catalog # : 41600-2	Aldrich Chemical Company Inc. PO Box 335 Milwaukee WI 53201 0335 Phone: 414 273 3850
26	Power supply	Part# B24G75 24 V DC power supply, 0.75A	Acopian Technical Co. PO Box 638 Easton, PA 18044 Phone: 800 523 9478
27	Pressure switch	Model # : 4NN-K4-N4-B1A-MM Range: 2-25psi	SOR Inc. 14685 W. 105th St. Lenexa, KS 66215 2003 Phone: 913 888 2630
28	Relay: plug/socket	Socket: Part# 7170K18 120V coil voltage plug: Part# 7170K14 24 V coil voltage plug: Part# 7170K12	Mc Master Carr PO Box 440 New Brunswick, NJ 08903-0440 Phone: 732 329 3200
29	SCR Power Controller	Firing card model# : 1021A Unit part# : 1651-24-20 20A, 240V, 1PH, 3 PAK SCR controller	Control Concepts 7870 Park Drive Chanhassen, MN 55317

Phone:	800 765	2799
I MOMO,	000 / 00	

30	Solenoid valve	Nomally open, part# : 8210G34 Normally closed, part# : 8210G94 Control voltage: 120 V AC 3/4" NPT female fittings	ASCO: Automatic switch Co. 50 Hanover Rd., Florham Park, NJ 07932 Phone: 800 937 ASCO
31	Strip heaters	Model# : OT-1505W 1kW strip htr. with Chrome steel sheath	Chromalox Leo Pelkus Inc. 170 Worcester St., PO Box 349 Wellesly Hills, MA 02181 Phone: 617 235 8040
32	Swagelok fittings	Pipe fittings: Connectors, adapters, reducing unions etc.	Cambridge Valve and Fitting Inc. 50 Manning Road, Billerica, MA 01821 Phone: 781 272 8270
33	Thermal paste	Omegatherm thermally conductive heat. transfer paste Part# : OT-201	Omega Engineering Inc. One Omega Drive PO Box 4047 Stamford, CT 06907-0047 Phone: 800-TC-OMEGA
34	Thermocouple wire, Mullite sheath	Type K thermocouple wire, 0.032" dia. Part# : A-13-21 Double bore Mullite thermocouple ID 0.060" insulator, part# : A-14-2 Used for temperature profiling	Nanmac Corporation 9, Mayhew Street Framingham, MA 01702 Phone: 800 758 3932
35	Process, overtemp. thermocouples	Part# : KQIN-14G, type K with Inconel sheath	Omega Engineering Inc. One Omega Drive PO Box 4047 Stamford, CT 06907-0047 Phone: 800-TC-OMEGA
36	Thermostat	Part# : OA C 606, Ceramic thermostat Type H, 250 V AC 5A max, NC, opens at 200 C	Selco Products Co. 709 N. Poplar St. Orange, CA 92868-1013 Phone: 714 712 6200
37	Transformer	1 kVA 240 V/120 V Model# : TA-2-81217	Acme Electric Corp. 4815 W. 5th Street Lumberton, NC 28358 Phone: 800 334 5214
38	Silicone tubing	Part# : LU- 06411-63 3/32" ID, 5/32" OD Silicone tubing to make seal at the Ultem-brass interface of the power feedthrough	Cole Parmer Instrument Co. E. Bunker Court, Vernon Hills IL 60061 Phone: 800 323 4340

39	Ultem fitting	Ultem stock purchased from Mc Master Carr. Fittings machined according to drawing in Appendix 2	Central Machine Shop, MIT		
40	Viewport	Part# : 450043 Model # : KVP-150 W 0.9" view diameter ports for use on furnace door	MDC Vacuum Products Corp. 23842 Cabot Blvd. Hayward, CA 94545 Phone: 510 265 3500		

Appendix 2: Furnace Part Drawings

Totals

Laboratory Vacuum Furnace

Lab for Manufacturing and Productivity, 3DP Project revised 5/25/99, D. Brancazio

			Furnace		
Table 1 (summa	ary)		Box	Door 1	Door 2
A	1/4" NPT		3	6	
В	3/8" NPT		7		
С	4-40 UNC-2B, .20 deep	steel insert	20	2	2
D	1/4-20 UNC-2B, .38 deep	steel insert	59	7	7
Е	8-32 UNC-2B, .25 deep	steel insert	16		
F	QF-40 Flange fitting		8		2
G	3/8-16 UNC-2B, .50 deep	steel insert	8		
н	f .255 +.005,000, .50 deep	flat bottom	24		
J	standoffs, 1" OD, 1/4" NPT end (detail F)		24		
к	Welded cups, 1.25 OD, .80 ID (detail H)		6		
L	** fitting TBD		1		
М	1/4-20 UNC-2B, .38 dp	steel insert		8	8
N	f .4062 thru			8	8
	welded pockets, dwg FUR003		31	4	4

Complete List of Features

F	uı	n	ac	е.	в	οх

Ν

8 f .4062 thru

1 unac	C DON			
	31	welded pockets, dwg FUR003		welded pockets for heaters, insulation
1	24	welded standoffs, 1" OD		power feedthroughs
L	1	** fitting TBD		pressure relief fitting
в	1	3/8" NPT		pressure relief fitting
Н	24	f .255 +.005,000, .50 deep	flat bottom	heater plate locators
к	6	Welded cups, 1.25 OD, .80 ID		hearth plate locators
В	6	3/8" NPT		gas ports
Α	3	1/4" NPT		thermocouples
F	2	QF-40 Flange fitting		piston up/down
F	1	QF-40 Flange fitting		debind outlet
F	1	QF-40 Flange fitting		part thermocoouples
F	l	QF-40 Flange fitting		infiltrant gate
F	1	QF-40 Flange fitting		vacuum port
F	2	QF-40 Flange fitting		extras
С	20	4-40 UNC-2B, .20 deep	steel insert	thermostats
D	40	1/4-20 UNC-2B, .38 deep	steel insert	strip heaters
D	8	1/4-20 UNC-2B, .38 deep	steel insert	Guard mounts
D	3	1/4-20 UNC-2B, .38 deep	steel insert	hinge mount
D	8	1/4-20 UNC-2B, .38 deep	steel insert	Mounting Bar mounts
G	8	3/8-16 UNC-2B, .50 deep	steel insert	Door bolts
Е	16	8-32 UNC-2B, .25 deep	steel insert	clamps for door
Door l	(with therm	ocouples)		
	4	welded pockets		insulation
A	6	1/4" NPT		thermocouples
м	8	1/4-20 UNC-2B, .38 dp	steel insert	handles
D	4	1/4-20 UNC-2B, .38 deep	steel insert	strip heaters
С	2	4-40 UNC-2B, .25 deep	steel insert	thermostats
D	3	1/4-20 UNC-2B, .38 deep	steel insert	hinge mount
N	8	f .4062 thru		bolt holes
Door 2	(with view	ports)		
	4	welded pockets		insulation
F	2	QF-40 Flange fitting		viewports
М	8	1/4-20 UNC-2B, .38 dp	steel insert	handles
D	4	1/4-20 UNC-2B, .38 deep	steel insert	strip heaters
С	2	4-40 UNC-2B, .25 deep	steel insert	thermostats
D	3	1/4-20 UNC-2B, .38 deep	steel insert	hinge mount

bolt holes

Laboratory V	acuum Fu	rnace - F	eature Size and Loo	cation Tabl	es		
Lab for Manu	facturing	and Produ	ctivity, 3DP Project				
revised 4/29/	99, D. Bra	incazio					
TABLE 2	(FRONT,	FUR001	Sheet 1)	TABLE 3	(RIGHT S	IDE, FUR	001 Sheet 1)
X, Y location	n toleranc	es: +/0	10	X, Y locati	on toleran	ces: +/0)10
	Х	Y			Х	Y	
E1	0.824	18.825		C1	17.536	16.9	
E2	1.425	19.426		C2	18.5	16.9	
E3	1.531	18.118		C3	17.536	5.25	
E4	2.132	18.719		C4	18.5	5.25	
E5	17.075	19.426		D1	4.75	15.65	
E6	17.676	18.825		D2	19	15.65	
E7	16.368	18.719		D3	4.75	6.5	
E8	16.969	18.118		D4	19	6.5	
E9	16.969	3.882		D5	11.875	16.9	
E10	17.676	3.175		D6	11.875	14.4	
E11	16.368	3.281		D7	11.875	7.75	
E12	17.075	2.574		D8	11.875	5.25	
E13	0.824	3.175		D9	1.75	21	
E14	1.531	3.882		D10	22.25	21	
E15	1.425	2.574		D11	1.75	1	
E16	2.132	3.281		D12	22.25	1	
G1	0.935	14.444		J1	7.31	19.21	
G2	5.806	19.315	**************************************	J2	15.56	19.21	
G3	12.694	19.315		J3	21.56	19.21	
G4	17.565	14.444		J4	7.31	12.21	
G5	17.565	7.556		J5	15.56	12.21	
G6	12.694	2.685		J6	21.56	12.21	
G7	5.806	2.685		J7	7.31	10.06	
G8	0.935	7.556		J8	15.56	10.06	
				J9	21.56	10.06	
				J10	7.31	3.06	
				J11	15.56	3.06	· · · · · · · · · · · · · · · · · · ·
				J12	21.56	3.06	

TABLE 4	(TOP VIE	W, FURO	01 Sheet 4)					
X, Y location tolerances: +/010								
	X	Y						
A1	14.25	13.25						
A2	14.25	15.25						
B1	3.75	2.5						
B2	14.75	2.5						
C1	4.25	18.5						
C2	4.25	17.536						
C3	14.25	18.5						
C4	14.25	17.536						
D1	3	4.75						
D2	3	19						
D3	15.5	4.75						
D4	15.5	19						
D5	1.75	11.875						
D6	4.25	11.875						
D7	14.25	11.875						
D8	16.75	11.875						
F1	5.25	6						
F2	5.25	15						
F3	9.25	7						
F4	9.25	12						
F5	9.25	18.5						
F6	13.25	10						

TABLE 5	(BACK, F	UR001 Sheet 6)	TABLE 6	(LEFT SI	DE, FURO	01 Sheet 6)	
X, Y location	n toleranc	es: +/010	X, Y locati	on toleran	ices: +/(010	
	Х	Y		Х	Y		
B1	3.75	19.75	C1	17.536	16.9		
B2	14.75	19.75	C2	18.5	16.9		
B3	3.75	2.75	C3	17.536	5.25		
B4	14.75	2.75	C4	18.5	5.25		
C1	14.911	15.25	D1	4.75	15.65		
C2	15.875	15.25	D2	19	15.65		
C3	14.911	6.25	D3	4.75	6.5		
C4	15.875	6.25	D4	19	6.5		
D1	2.125	16.5	D5	11.875	16.9		
D2	16.375	16.5	D6	11.875	14.4		
D3	2.125	5	D7	11.875	7.75		
D4	16.375	5	D8	11.875	5.25		
D5	9.25	17.75	D9	1.75	21		
D6	9.25	15.25	D10	22.25	21		
D7	9.25	6.25	D11	1.75	1		
D8	9.25	3.75	D12	22.25	1		
L1	9.25	11.75	D13	1.17	12.75		
			D14	1.17	11		
			D15	1.17	9.25		
			J1	7.31	19.21		
			J2	15.56	19.21		
			J3	21.56	19.21		
			J4	7.31	12.21		
			J5	15.56	12.21		
			J6	21.56	12.21		
			J7	7.31	10.06		
		· · · · · · · · · · · · · · · · · · ·	J8	15.56	10.06		
			J9	21.56	10.06		
			J10	7.31	3.06		
			J11	15.56	3.06		
			J12	21.56	3.06		

TABLE 7	(SECTION E-E, FUR001 Sheet 7)			TABLE 8 (BOTTOM, FUR001 Sheet 7)				
X, Y location tolerances: +/010				X, Y location tolerances: +/010				
	Х	Y			Х	Y		
H1	2.08	3.25		A1	4.25	13.25		
H2	2.08	6.25		B1	9.25	16.75		
H3	2.08	10.375		C1	13	3.786		
H4	2.08	13.375		C2	13	4.75		
H5	2.08	17.5		C3	5.5	3.786		
H6	2.08	20.5		C4	5.5	4.75		
H7	16.42	3.25		D1	14.25	4.75		1
H8	16.42	6.25		D2	4.25	4.75		
H9	16.42	10.375		D3	14.25	19		
H10	16.42	13.375		D4	4.25	19		
H11	16.42	17.5		D5	15.5	11.875		
H12	16.42	20.5	••••••••••••••••••••••••••••••••••••••	D6	13	11.875		
H13	2.58	3.75		D7	5.5	11.875		
H14	2.58	5.75		D8	3	11.875		
H15	2.58	9.875		D9	17.25	1.5		
H16	2.58	13.875		D10	17.25	2.5		
H17	2.58	18		D11	17.25	21.5		
H18	2.58	20		D12	17.25	22.5		
H19	15.92	3.75		D13	1.25	1.5		
H20	15.92	5.75		D14	1.25	2.5		
H21	15.92	9.875		D15	1.25	21.5		
H22	15.92	13.875		D16	1.25	22.5		
H23	15.92	18		F1	9.25	21	!	
H24	15.92	20		F2	9.25	12		
				K1	12	7		
				K2	6.5	7		
				K3	12	13		
				K4	6.5	13		
				K5	12	19		
				K6	6.5	19		

TABLE 9	(FRONT,	FUR002 \$	Sheet 1)				r -		
X, Y location tolerances: +/010									
	X	Y							
A1	-3.00	4.00	******************************						
A2	0.00	4.00							
A3	-3.00	0.00							
A4	0.00	0.00							
A5	-3.00	-4.00							
A6	0.00	-4.00					ļ. 1		
C1	3.5	6.482							
C2	3.5	5.518							
D1	2.00	7.125							
D2	2.00	-7.125							
D3	0.75	0.00							
D4	3.25	0.00							
D5	-8.715	2.375							
D6	-7.465	0.625							
D7	-8.715	-1.125							
M1	-6.00	3.25							
M2	-4.625	3.25							
M3	4.625	3.25							
M4	6.00	3.25							
M5	-6.00	-3.25							
M6	-4.625	-3.25							
M7	4.625	-3.25							
M8	6.00	-3.25							
N1	8.315	3.444							
N2	3.444	8.315							
<u>N3</u>	-3.444	8.315							
N4	-8.315	3.444							
N5	-8.315	-3.444							
N6	-3.444	-8.315							
N7	3.444	-8.315							
N8	8.315	-3.444							

TABLE 10 (FRONT, FUR010 Sheet1)								
X, Y location tolerances: +/010								
	Х	Y						
C1	3.5	6.482						
C2	3.5	5.518						
D1	2	7.125						
D2	2	-7.125						
D3	0.75	0						
D4	3.25	0						
D5	-8.715	2.375						
D6	-7.465	0.625						
D7	-8.715	-1.125						
F1	-1	2.5						
F2	-1	6.25						
M1	-6	3.25						
M2	-4.625	3.25						
M3	4.625	3.25						
M4	6	3.25						
M5	-6	-3.25						· · · · ·
M6	-4.625	-3.25						
M7	4.625	-3.25						
M8	6	-3.25						
N1	8.315	3.444						
N2	3.444	8.315						
N3	-3.444	8.315						
N4	-8.315	3.444						
N5	-8.315	-3.444						
N6	-3.444	-8.315						
N7	3.444	-8.315						
N8	8.315	-3.444						



























- - 2





n de la la Antes de la composition



1. (20. ¹2¹¹)



i i stati















dia di Mana



and strand the






tipe:





in distyledation

















1999年1月1日(1993年) 1999年(1993年) 1999年(1993年)



Note: All tolerances +-0.005", unless

otherwise stated

Drawing #FUR025				
Ultem power feed through	All			
Scale: 4/1	dimensions in inches			



Note: All tolerances +-0.005" unless otherwise stated

Drawing #FUR026					
Brass electrical connector	All				
Scale: 4/1	dimensions in inches				



Appendix 3: Sample Program for the Large Scale Furnace

The controller can store up to 49 different programs in memory. After choosing a program number that the user wants to program into, pressing the "func" and the "prog" keys on the controller panel enables the user to write into the program. A typical program that was used in a furnace run to sinter SS420 powder is presented here. This program was used to carry out run 5 that is described in Chapter 7.

Segment #	Set point	Time	Event 2	Event 3	Event 4	Event 5	PID #
1	10	5 sec					1
2	10	1 min	on	on	on	on	1
3	170	2 min	on	on	on	on	1
4	170	120 min	on	on	on	on	1
5	200	1 min	on	on			1
6	500	60 min	on	on			1
7	1000	100 min	on	on			2
8	1200	40 min	on	on			3
9	1200	60 min	on	on			3
10	20	10 min	on				3
11	20	480 min	on				3

Notes:

1. Segments 1 and 2 are dummy segments. If the very first segment of the program has the contents of segment 3, i.e. SP = 170 C, Time = 2 min, the controller interprets this to mean that it has to stay at 170 C for 2 minutes, and NOT as " ramp to 170 C in 2 minutes". However, the presence of dummy segment 2 before segment 3 causes the furnace to ramp to 170 C in 2 minutes starting at the end of segment 2 as desired.

2. Event 2 turns on the flow of Forming gas (and shuts off the flow of Argon), events 3 and 4 turn banks 1 and 2 respectively of external heaters on. Event 5 turns the top six internal heaters off.

3. Event 5 is on in segments 2 to 4. This causes power to be supplied only to the bottom zone of internal heaters. Also, the set point in segments 3 and 4 was chosen to be 170 C and not 200 C, which is the recommended purging temperature. This was done in order to account for the lag between the control thermocouple that is located in the top heater zone and the bottom heaters that are heating the furnace. This lag causes the bottom zone to overheat above the set point. It was found that a set point of 170 C proved right for the steady state furnace temperature during the purge to settle at around 200 C.

4. Note that segment 4 has been programmed to be 2 hours long. However, during the actual run, the "hold" key on the controller was pressed while the furnace run was in this segment. Once the dew point in the furnace got to below -25 C (which happened around 3 hours after the beginning of this segment in run 5 discussed in Chapter 7), the program was advanced to segment 5 by pressing the "prog" and "disp" keys on the controller together.

5. From segment 5 onwards, bank 2 of the external heaters are shut off and both zones of internal heaters are turned on (event 5 is off).

6. The ramp to 1200 C occurs at a rate of 5 C/min. Note that the ramp has been split into three sections, each of which has a different PID variable for control. PID1 is used for segments with a set point (SP) <= 500 C, PID2 for segments with 500 C< SP<=1000 C and PID 3 for segments with SP>1000 C.

7. The furnace is held at 1200 C for 60 minutes, during which time the SS420 powder sinters.

8. In segment 10, the furnace has been set to cool from 1200 C to 20 C in 10 minutes. Obviously, this is not possible (or desirable) since the only way the part can cool is by losing heat by natural convection. Thus natural convection limits the cooling rate of the furnace.

9. In segment 11, gas flow has been programmed to be left on for 8 hours after the furnace starts to cool down. Eight hours is enough time for the furnace to cool down to a temperature (< 250 C) at which there is no risk of oxidizing the SS part if flow of Forming gas is turned off.

Appendix 4: Using a Nextel Lined SS Crucible for Infiltrating Parts

In the course of performing experiments on directional solidification (Chapter 4), it was found that Graphite crucibles used for melting the infiltrant often used to crack due to thermal expansion induced stresses. Also, the Boron nitride coating that was used to coat the inside of a Graphite crucible would sometimes crack and cause the infiltrant to stick to the Graphite, rendering the crucible useless for future use.

Ideally, one would want to use a SS crucible (which would be much less likely to crack than Graphite) to melt the infiltrant. However, the only issue is that the infiltrant (typically Copper or Bronze) sticks to SS. Nextel, the fabric that was used extensively in the large scale furnace (see section 5.3.3 of Chapter 5), has the property that Copper or Bronze do not wet it. Thus, it was decided to line the inside of a SS crucible with Nextel BF20 fabric (see Appendix 1 for vendor listing) in order to carry out infiltrations. Molten infiltrant does not leak through the Nextel lining, which acts like a bag and contains the melt within itself. It is important to use unsized Nextel (sizing is a chemical coating applied to Nextel fabric to make sewing easy). Otherwise, vapor generated when the sizing burns off during a furnace run will contaminate the furnace atmosphere.

The picture below shows a Nextel lined SS crucible that was used to infiltrate parts in the large scale furnace (the part shown in Figure 7.8 of Chapter 7 was infiltrated using this crucible).



References

1. CRC Handbook of Chemistry and Physics, 80th ed., CRC Press, New York, 1999.

2. German, R.: Powder Metallurgy Science, 2nd ed., Metal Powder Industries Federation, Princeton, NJ, 1994.

3. Guo, Honglin.: " Alloy Design for Three- Dimensional Printing of Hardenable Tool Materials", Ph.D. Thesis, Massachusetts Institute of Technology, 1998.

4. Lorenz, A. M.: "Economical Furnace Processing for 3-D Printed Metal Parts", SM Thesis, Massachusetts Institute of Technology, 1998.

5. Mark's Standard Handbook for Mechanical Engineers, 10th ed., McGraw-Hill, Inc., New York, 1996.

6. Mills, A. F.: Heat Transfer, 2nd ed., Prentice Hall, NJ, 1999.

7. Shames, I. H.: Mechanics of Fluids, 3rd ed., McGraw-Hill, Inc., New York, 1992.

8. Strehlow, R. A.: Combustion Fundamentals, McGraw-Hill, Inc., New York, 1984.